Actual					Variance	Forecast	Budget	Variance
	Budger	Variance	Actual	Budget	a a manica			monute of the second of the second
		1121025		10 000		21.522		2012 2012
								22,96
								2,38
								-23,30
								-37,26
								6,80
0	0	0	0	0	0	0	0	
	1							
7,586	0		33,773	0			0	30,00
2,436		2,436						60,00
78,092	88,406	-10,313	1,009,974	1,030,592	-20,618	1,118,998	1,118,998	
7,653	8,664	-1.011	98,977	100,998	-2,021	109,662	109,662	
11,394	12,898	-1,505	147,355	150,363	-3,008	163,262	163,262	
0	0	0	0	0	0	0	0	
1,952	2,210	-258	25,249	25,765	-515		27,975	
0	0	0	0	0	0		0	
		-1,871						
0	0	0	0	and the second se				
				4,844				
				100 110				
					-3,870		210,036	Lane of Maline
					2 452		133 049	
78,092					-20,618			
			11-11-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-			1		
1 .1 840	AAEEA	40 984	1 022 025	1 099 847	56 012	1 112 974	1 191 201	-18,22
								-120,56
								-160,14
								123.02
				45,766	-26,009			-27,72
					-1,021,566	3,277,811	4,255,797	-977,98
12,580			142,936	142,936	0	164,937	164,937	
181,847	1,112	180,735	1,259,099	2,468,428	-1,209,329			-1,010,35
11,154	11,154	0	18,103	18,102	. 1			
33,911	111,594							-063,27
								-1,770,09
								490,15
								-584,99
								-155,08
1,275,086	1,318,227	-45,140	21,205,913	25,172,175	-3,900,200	25,755,002	20,000,079	-4,010,21
	1					Spending	Secured	
								Variance
			1	1	E PULLER S	CLO STORAGE		
						3,964,557		-18,22
	Contraction Victorian Contraction	and the state of		1	AND A COMPANY AND AND A			-120,56
								-160,14
					u <u>m</u> an			
								20.00
					1 12 Marsh 1			-690,90
			and the second		and the second			498,64
								499,64
	-							
-						5,042,000	0.042.000	
						8,515 150	6.025.000	490,15
								-584,99
-						3,500,000		
							45,856,774	-588,03
	7,586 2,436 78,092 7,653 11,394 11,394 11,392 0 14,166 0 367 0 14,658 0,332 0,285 78,092 4,810 270,108 13,774 6,045 2756,393 12,580 181,847 11,154 33,911 12,580 181,847 11,154 33,911	49,414         73,839           560         0           -10,769         7,533           8,163         3,083           474         1,350           0         0           7,586         0           7,586         0           7,653         8,664           11,394         12,898           0         0           7,653         8,664           11,394         12,898           0         0           0         0           1,952         2,210           0         0           0         0           0         0           0         0           0         0           0         0           367         416           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0		49,414         73,839         -24,425         686,522           550         0         550         2935           -10,759         7,533         -18,292         50,117           8,163         3,083         5,080         72,483           474         1,350         -876         20,505           0         0         0         0         0           7,586         0         7,586         33,773           2,436         60,042         78,092         88,406         -10,313         1,009,974           7,653         8,664         -1,011         98,977         11,394         12,898         -1,605         147,355           0         0         0         0         0         0         0           1,952         2,210         -258         25,249         0         0         0           0         0         0         0         0         0         0         0           14,166         16,037         -1,871         183,209         0         0         0           0         0         0         0         0         0         0           0         0         0	$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	49,414         73,839         -24,426         686,522         799,359         -112,837         811,620           -10,759         7,533         -18,292         50,117         82,867         -32,749         67,100           8,183         3,083         5,080         72,483         84,917         -12,434         50,710           474         1,350         -870         20,505         14,850         5,665         23,000           0         0         0         0         0         0         0         0         0         0           7,586         0         7,566         33,773         0         33,773         30,000           7,653         86,646         -1,011         98,977         100,998         -2,021         109,682           111,344         12,898         -1,505         147,955         150,333         -3,009         169,202           10         0         0         0         0         0         0         0         0           14,160         16,037         -1,871         183,209         186,949         -3,740         202,986           0         0         0         0         0         0         0 <td< td=""><td>48.414         73.838         -24.426         688.622         706.358         -1.12.837         811.620         873.188           1.6.759         7.633         -18.292         50.117         82.867         -32.749         67.100         90.400           4.6         3.063         5.060         72.848         64.60         -1.246         67.100         90.400           4.6         3.060         72.468         64.60         -1.246         60.700         90.90           7.686         0         7.586         0         7.586         0.021         60.000         0         0           7.6802         86.406         -10.313         1.0008.074         1.030.552         -20.012         109.662         108.662           7.653         8.664         -1.011         98.977         100.968         -2.021         109.662         109.662         108.662           1.962         2.210         -2668         25.766         -615         27.976         27.976         27.976           1.962         2.210         -1.871         183.209         186.940         -3.740         20.62.986         20.62.986           0         0         0         0         0         0         <t< td=""></t<></td></td<>	48.414         73.838         -24.426         688.622         706.358         -1.12.837         811.620         873.188           1.6.759         7.633         -18.292         50.117         82.867         -32.749         67.100         90.400           4.6         3.063         5.060         72.848         64.60         -1.246         67.100         90.400           4.6         3.060         72.468         64.60         -1.246         60.700         90.90           7.686         0         7.586         0         7.586         0.021         60.000         0         0           7.6802         86.406         -10.313         1.0008.074         1.030.552         -20.012         109.662         108.662           7.653         8.664         -1.011         98.977         100.968         -2.021         109.662         109.662         108.662           1.962         2.210         -2668         25.766         -615         27.976         27.976         27.976           1.962         2.210         -1.871         183.209         186.940         -3.740         20.62.986         20.62.986           0         0         0         0         0         0 <t< td=""></t<>

	31/03/2004	30/04/2004	31/05/2004	30/06/2004	31/07/2004	31/08/2004	30/09/2004	31/10/2004	30/11/2004	31/12/2004	31/01/2005	29/2/2005
FIXED ASSETS	34,090	35,800	36,252	39,774	98,473	97,122	94,634	93,863	91,375	92,640	96,784	103,606
	34,090	35,800	36,252	39,774	98,473	97,122	94,634	93,863	91,375	92,640	96,784	103,606
CURRENT ASSETS												and the second s
Trade Debtors	2,003,455	3,221,220	3,404,964	3,083,030	3,082,234	5,188,900	5,357,348	5,385,325	7,553,865	5,551,588	4,655,862	954,697
Other Debtors	5,774	4,282	4,282	4,425	4,425	4,425	4,425	-1,575	-1,575	-3,427	-3,457	0
Prepayments & Accrued Income	20,788	20,304	20,009	1,178	883	0	0	0	0	0	0	2,317,809
CEC Loan	0	0	0	0	0	0	0	0	0	0	0	0
Petty Cash	424	62	112	69	319	25	18	48	48	48	68	68
	2,030,441	3,245,868	3,429,367	3,088,702	3,087,860	5,193,350	5,361,791	5,383,798	7,552,338	5,548,209	4,652,473	3,272,574
CURRENT LIABILITIES						1.00						1.0
Trade Creditors	1,925,102	1,251,205	1,388,699	1,862,376	2,460,584	2,195,592	1,712,746	2,514,223	1,894,261	2,085,329	2,810,521	1,582,802
Employee Creditor	-209	577	523	53	721	169	32	40	1,437	586	-257	-227
Bank Account	-229,479	1,218,285	1,102,852	405,612	-46,864	2,326,045	1,895,795	1,637,198	3,081,926	349,478	-261,408	-766,218
Pension Creditor	11,985	12,615	13,245	10,546	10,598	9,973	10,540	11,157	11,726	12,032	12,337	11,464
Capital Grant	34,090	34,090	34,090	34,090	34,090	34,090	34,090	34,090	34,090	34,090	34,090	34,090
Accruais	239,858	715,738	854,104	750,694	670,642	654,870	1,707,197	1,182,709	2,497,047	3,007,556	2,070,544	2,366,512
VAT Payable/(Refundable)	56,514	19,465	38,960	29,879	18,870	32,401	56,643	59,754	82,307	112,018	44,734	63,307
PAYE/NIC	25,670	28,667	32,095	34,227	36,692	35,178	37,236	37,191	38,773	37,614	36,550	82,303
Corporation Tax	0	0	0	0	0	0	0	0	0	0	0	0
Other Creditors	0	26	52	0	0	1,156	1,146	299	1,146	1,146	1,146	1,146
	2,063,531	3,280,668	3,464,619	3,127,475	3,185,333	5,289,472	5,455,425	5,476,660	7,642,713	5,639,849	4,748,257	3,375,180
NET CURRENT ASSETS/(LIABILITIES)	-33,090	-34,800	-35,252	-38,774	-97,473	-96,122	-93,634	-92,862	-90,375	-91,640	-95,784	-102,606
Liabilities > 1 Year	0	0	Ō	0	0	0	0	0	0	0	0	0
NET ASSETS	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000
Represented by:												
Share Capital	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000
Reserves	0	0	0	0	0	0	0	0	0	0	0	0
Profit & Loss Account	0	0	Ö	0	0	0	0	0	0	0	0	0
Balance as at Period End	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000

## 6 Balance Sheet – Month End and Year to Date Progress

7 Cash Flow - Year	to Date	and Fore	cast												
Feb-05	Ave a	-													
100.00								ACTUAL		101				FORECAST	
			Apr-04	May-04	Jun-04	Jul-04	Aug-04	Sep-04	Oct-04	Nov-04	Dec-04	Jan-05	Feb-05	Mar-05	Total
			Apr-04	may-04	Jun-04	Jui-04	Aug-04	Sep-04	001-04	1404-04	Dec-04	Jan-05	rep-us	mai-03	TUtar
Balance b/forward			229 478 91	-1 218 284 50	-1,102,852.00	-405 611 62	46,864,12	-2,326,044.50	-1 895 794 49	-1.637.197.41	-3.081.925.71	-349.477.40	261,408.01	765,931.44	229,478.9
		near the state				1001011104									
Income	0	1		a destruction of the	all and the second	Sec. Sec. 1		a contractor		The second second	100 C	denter march			
Sales Ledger					2,551,626.00			2,447,497.79			5,156,234.25				24,179,922.3
Miscellaneous			10.00	3.00	3,795.39	9,810.64	943.74	2,163.20	6,200.00	0.00	3,637.65		38.65	0.00	29,842.2
	-		161100	1,106,000,60	£1000,46 1100	2,140,010.01	£,0(£.40	2,440,000.00	6,100,011.66	002,000.01	0,100,071.00	2,702,002.01	2,700,020,00	000,200.00	E-11200110110
Expenditure				1111	and the second s		5 Part 10							and so the	1
Purchase Ledger			1,297,114.74	1,537,368.70	1,750,688.04		2,266,112.03				2,290,801.50				22,700,137.4
Expenses Ledger			7,000.00	123.10	0.00	0.00	1,144.36	552.69	216.64	1,174.46	1,956.28	842.31	3,104.40	0.00	16,114.2
Miscellaneous			143,776.17	109,440.98	107,492.97	173,788.30	108,524.66	128,895.93						167,427.80	
			1,447,890.91	1,646,932.78	1,858,181.01	1,694,440.77	2,375,781.05	2,019,410.98	2,531,720.14	2,307,537.11	2,427,423.59	2,121,197.20	2,284,499.66	1,750,002.80	24,465,018.0
Net Movement in Month			-1,447,763.41	115,432.50	697,240.38	452,475.74	-2,372,908.62	430,250.01	258,597.08	-1,444,728.30	2,732,448.31	610,885.41	504,523.43	+791,705.94	-255,253.4
Balance c/forward			-1,218,284.50	-1,102,852.00	-405,611.62	46,864.12	-2,326,044.50	-1,895,794.49	-1,637,197.41	-3,081,925.71	-349,477.40	261,408.01	765,931.44	-25,774.50	-25,774.5
NEXT MONTH FORECAST:	Assumptions	5										1			
	Set a substantia substantia substantia.														
Income						_									
Sales Ledger					-									045 000 04	-
Invoices issued to CEC in Febr	all a second and a second as													915,032.21	
Invoices issued to Clacks in Fe												_		25,058.52	
Involces issued to FETA in Febr		Lanar												13,089.50	
Invoices issued to Scottish Parl Invoices issued to SMI in Octob		ruary			and the second second second	and a company of the second	Community of the				1			280.00	
Involces issued to One-Ticket in		Paid 11/3/05	1100000	i in the second s										4,491.91	
Invoices issued to One-ricket in	reordary	Paid 11/3/05											-	958,296.86	
Accrued Income (Month End Ac	cruals), Fixed /	Asset Adjustme	nts etc.											2,317,809.00	
														3,276,105.86	1
Expenditure									-	-		-			
Trade Creditors per Balance S	theat	111111		FT THEFT									3 - Walter	1,582,802.00	South
Employee Creditor per Balance			111.11											-227.00	
Aged Creditors Li		-												1,582,575.00	
Miscellaneous	artig contenuo														
Pension Fund(s) - Contribution	s Due on 19/3	05									1			11,464.17	
HM C&E - VAT Return to 31/3/05														0.00	
PAYE/NI - Due on 19/3/05	te do ior pays		-								1			82,302.80	
March Payroll - 28 members of	staff										10			85,000.00	
Bank Interest - Quarter ending											i inga ana			0.00	
Bank Charges for month	and the second		No. of Concession, Name								and the second			100.00	
Petty Cash for month	and the second s													25.00	
PRODUCT STATE OF COLUMN		- I - magain												167,427.80	
Accrued Expenditure.													-	2.366.511.91	
Capital Grant/Fixed Asset Purch	ase etc.	S								-	1 1 1 1			34,090.41	
a selection and the selection of the sel	Total of the start					Accruals per			G					2,400,602.32	-



Agenda Item 5c

# Finance

c) Draft tie Business Plan\*



Agenda Item 5c

## Finance

c) Draft tie Business Plan\*



tie limited

Business Plan - 2005/06

## **DRAFT V7**

15th March 2005

## CONTENTS

1

Executive Summary

Detailed project programmes are available for inspection which support the project plans.

### 1 Executive Summary

Edinburgh has one of the fastest growing economies of any major city in Europe and is facing significant expansion in employment and housing. In order to cope with this, a number of transport schemes have been identified in Edinburgh's Local Transport Strategy. These fall within a comprehensive programme of related schemes collectively labelled the Integrated Transport Initiative for Edinburgh and South East Scotland (the "ITI"). In addition to the Edinburgh-based schemes, **tie** is charged with delivering two key heavy rail projects for the Scottish Executive. The time-scale for delivery of these schemes ranges over many years ahead.

**tie** is a private limited company wholly owned by the City of Edinburgh Council (CEC). The operating agreement between **tie** and CEC requires **tie** to submit a business plan in draft to CEC in this case for the year to 31 March 2006 (FY06).

During 2004/5 to date, tie has managed the following projects:

- Development and procurement of the Congestion Charging (CC) scheme;
- Tram schemes parliamentary process supporting two lines and preparation of the Bill for the 3<sup>rd</sup> line, early tram operator involvement, service integration and commencement of system procurement;
- Construction of Edinburgh Fastlink (formerly WEBS);
- Construction of Park and Ride site at Ingliston;
- Preparation of the Bill and scheme development for the Edinburgh Airport Rail Link (EARL);
- Planning for procurement and construction of the Stirling/Alloa/Kincardine railway (SAK); and
- "One Ticket" public transport integrated ticketing system.

Total outturn expenditure in the 04/05 financial year (hereinafter referred to as FY05) is forecast at £24m.

WEBS and the Ingliston Park and Ride will be largely completed this year, although **tie** will have some continuing responsibilities. The remaining projects will form the core programme for the next three years and represent the basis of this business plan

Section 2 of this plan describes **tie**'s Corporate Governance processes and Section 3 sets out **tie**'s approach to financial management. Section 4 describes the progress made on the projects for which **tie** has responsibility. Section 5 describes the initiatives undertaken by **tie** to develop and strengthen its internal processes. Section 6 provides a detailed analysis of the resource requirements, planned expenditure and sources of funding for **tie**'s projects. The appendices provide further detail.

Of the outturn forecast for FY05 of £24m, some £10m relates to actual construction of the WEBS and Ingliston projects. The majority of the balance of expenditure relates to the development and procurement of the congestion charging, tram and EARL projects.

tie's internal costs (people and overhead) are c12% of total expenditure and are allocated to projects on an equitable basis.

The plan for FY06 is prepared on the following basis:

#### Congestion charging scheme

This scheme has been terminated

#### Tram project – parliamentary process

The actual expenditure will depend upon a number of variables which will drive the depth of work required by **tie**, principally the requirements of the Committees of MSPs and the need to respond to objectors. This is presently estimated at £3.4m for both lines 1 and 2, although as explained in the plan this is a highly subjective estimate. It is assumed some limited work on Line 3 is continued but this ceases in Q1 FY06

#### Tram Outline Business Case and Implementation

Project implementation stage has now commenced following the grant award earlier this year. This programme is initially focussed on the development of the Outline Business Case ("OBC") which encompasses the involvement of Transdev, the procurement strategy and tender process, service integration planning and detailed system design. The total costs in FY06 are estimated at £21.9m, allowing for some slippage of FY05 spend into FY06, £2m of expenditure on third party contracts, the estimated £1.2m cost of detailed revenue forecasting in relation to a fully integrated bus and tram system and a placemarker sum of £3m relating to advance utility diversion work. Further detail is provided in Section 4. The Executive anticipate that approval for this funding will be confirmed as part of the evaluation of the Outline Business Case in spring 2005, dependent on the demonstration of tangible progress. A dialogue is underway between the Council and the Executive on funding contribution by the Council toward this activity in FY06.

This plan does not address full-scale advance construction costs, which are under discussion with the Executive. These costs include the investment required to plan work on utility diversions, Network Rail interfaces, land and property acquisition; followed by spend on actual operational work in these areas. The timing and quantum of this expenditure will drive the overall programme. It is anticipated that all such work would be funded by the Executive as part of the overall construction cost. This will be assessed as part of the planned preparation of the OBC in Spring 2005.

#### EARL

Support to parliamentary process of the EARL Bill is estimated at £2.3m and a further sum of £3.3m is proposed for procurement and enabling works related to the project to accommodate stakeholder requirements and to maintain programme but which is not directly related to the parliamentary process. As with the Tram Bills, the actual parliamentary expenditure will depend upon the requirements of the Scottish Parliament. **tie** is in discussions with the Executive about the EARL Promoter role. No costs have been reflected for this role in the Plan as the scope remains uncertain.

#### SAK

SAK is a key project, but as presently structured will not involve **tie** in significant funding issues since these flow from Scottish Executive to the Promoter.

### FETA - Development of A Charging Order

tie submitted a proposal to the Forth Estuary Transport Authority (FETA) to assist them with the implementation of a Charging Order for the bridge.

tie will project manage the process and will assist FETA staff, including the Treasurer, by preparing briefs and contract documents as appropriate for the external services covering legal, financial and technical advice. External consultants would be engaged by FETA directly and tie would project manage and certify payments in accordance with the contract documentation. tie may also offer to use its own resources and consultants where specialised services were required and where insufficient time was available for procurement.

The personnel involved are existing members of staff and will normally be based in **tie**'s office for the duration of the commission.

Progress on **tie**'s projects in FY05 is set out in Section 4, together with a description of the challenges faced by each project in FY06 and the outline funding requirements. **tie** has skills in UK and European procurement, programme and project management, financial structuring and management and the management of specialist advisers on transport planning, project finance, engineering and legal issues. These skills will be brought to bear in predicting and controlling expenditure in the most cost-effective way.

tie has a unique position as a company with public sector ownership, ethos and objectives but a private sector approach to delivery. tie is not established for profit, and its goals are completely aligned with its public sector stakeholders in the delivery of transport projects. Since tie was established in April 2002, considerable progress has been made on these large-scale, complex and high profile projects against a background of new and developing parliamentary requirements. The proposals in this plan represent tie's best estimate of the resources required to maintain the momentum on these projects.

#### Principal matters which will follow on from Plan approval

There are a number of key issues which will require to be addressed prior to the commencement of the new financial year or which will be addressed as the year proceeds.

- Ratification by tie Board of Corporate Objectives set out in Section 5 of this Plan and approval of the Plan by the Board.
- Final determination of plans to recover value from the investment in the ITI Business System.
- Agreement from Scottish Executive on funding to continue Tram Lines 1 & 2 parliamentary process.
- Final determination of Tram Line 3 Bill.
- Development of the tram project OBC and dialogue with the Executive on funding support. This will include Line 1 & 2 advance construction work programme and expenditure.
- Submission date for EARL Bill to be finalised, potential subsequent re-programming, expenditure re-profiling and new funding confirmed.
- tie's role as Promoter of EARL Bill and funding requirement to be determined.



Agenda Item 6

# **Heavy Rail**

- a) EARL Project and Financial Progress Report
- b) EARL- Promoter Role
- c) SAK Project and Financial Progress Report



Agenda Item 6a&b

# **Heavy Rail**

- a) EARL Project and Financial Progress Report
- b) EARL- Promoter Role



Paper to:tie Board, 21st March 2005Subject:Commercial & in ConfidenceSubject:Edinburgh Airport Rail Link<br/>(Project Manager - Susan Clark)

From: Paul Prescott

Date: 16<sup>th</sup> March 2005

## Project Governance

Work is ongoing to confirm **tie** as the Promoter for EARL. A separate paper will be tabled at the meeting outlining progress and action and decisions required.

## **Bill Progress**

Good progress is being made with the Bill Documents. Skeletons are already in place for the Bill itself and the Explanatory Notes, with drafts due at the end of March. Drafts of maps, plans & sections are also in place, as are first drafts of the Promoter's Memorandum and Statement. A draft Environmental Statement is being reviewed within **tie**.

We have written to SE to enquire about the ability of the Private Bills Unit to process another Bill, as four are already listed as being in the system. The letter of response confirms that we should proceed with introduction of the Bill during June.

## **Costs & Preliminary Financial Case**

The cost report has now been produced, and indicates that costs are close to (and within) the previous SKM costs. The STAG appraisal is now underway to ensure that the business case still stands. The Preliminary Financial Case is being developed in parallel.

## Planning

We are in discussion with West Lothian Council to agree the scope for the timetabling exercise for Winchburgh Station. WLC are seeking developer funding for this.

Feedback from TRANSCO in relation to the diversion of the high pressure pipeline has indicated that the previous route can be altered to accommodate EARL.

## 3<sup>rd</sup> Parties

Assurance protocols being developed with both BAA & NR. Protective provisions are also being discussed with NR to prevent an objection to the Bill. A Basic Services Agreement extension with NR has been agreed and signed.

SC/PGP 16.03.05

<b>Report for Month Ending:</b>	28 Feb-05	ject Manager:	Susan Clar	-k		
		110			ĸ	
Start Date:		and the second	End Date:			
<b>Overall Progress Status</b>	Expen	diture 2004/5	1	Project	Life Funding	
	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	77%			116%	
Progress Key:				Finance Key	:	
On track for successful completion as p	programmed.			Within 10% of estimate		
Issues have arisen which may delay cor	npletion or require discussion/dir	rection.		10 - 20% out	side estimate	
Issues have arisen which will delay con				>20% outside	e estimate	
Critical Path / Milestone Items	Original Sta Date	rt Original Completion	Revised Completion	Progress (NS,IP,C)	Progress Status (G,Y,R)	
1. Cost Report	9-Dec-04	9-Dec-04	28-Feb-05	IP		
2. STAG Report	18-Feb	18-Feb	04-Apr	IP		
3. Finalise ES	03-Mar-05	03-Mar-05	07-Apr-05	IP		
4. Submit Bill	10-Mar-05	10-Mar-05	20-May-04	IP		

Service - Service - Service -	Funding	Budget	Original Cost Estimate	Start of Year Cost Estimate	Current Forecast	Variance
Previous Years	£744,204	£744,204	£744,204	£744,204	£744,204	£0
2004/5	£4,255,796	£4,255,796	£4,255,796	£4,255,796	£3,277,811	£977,985
2005/6	£0	£0	£0	£0	£287,085	-£287,085
2006/7	£0	£0	£0	£0	£0	£0
Future Years	£0	£0	£0	£0	£0	£0
Total for Project Life Cycle	£5,000,000	£5,000,000	£5,000,000	£5,000,000	£4,309,100	£690,900





#### Technical, Environmental

Cost report produced in costs in line with original SKM costs. First draft of Environmental Statement delivered and being reviewed by a number of statutory bodies. First deliverables from STAG delivered and audit ongoing by WSP.

#### Bill Process

Decision on Promoter still not finally resolved and this may impact the submission date of 9<sup>th</sup> May. Promoter's Memorandum & Statement progressing well. <u>PFC</u>

Work now started in earnest to produce PFC now that information being delivered from STAG and cost report. Meetings being arranged with SE to review both STAG and PFC to ensure they are fit for purpose prior to submission of the Bill.

#### Agreements

Agreement reached with NR to roll forward with existing Basic Services Agreement pending introduction of a new suite of agreements between NR/SE. Discussions are ongoing about inclusion of protective provisions in the Bill to prevent an objection. Negotiations have commenced with Baa re funding contribution.

09/03/2005

"I confirm that this report provides an accurate overview of the project progress and finance."

Project Manager's signature:

Date: 10305

Project Director's signature:

Date:

09/03/2005



Agenda Item 6b

# **Heavy Rail**

## b) EARL- Promoter Role

### Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

#### EARL BILL PROMOTION

#### Advice to tie Limited

#### Introduction

This note reviews the position reached as at 11 March 2005. We are aware that discussions have taken place between **tie** and DEC and **tie** and the Scottish Executive concerning matters covered in our earlier February and March 2005 advice notes to **tie**. We have been asked by **tie** Project Management to focus on the option involving **tie** as Promoter of the EARL bill and to identify risks and mitigation measures. We have advised **tie** on these matters in June and August 2004.

#### Scottish Executive ("SE") Position on Promotion

- 1. The SE position can be summarised as follows:
  - SE is focused on the short term issue of formally identifying a Promoter for EARL;
  - SE is not able to give any assurance that **tie** will continue to implement EARL as Authorised Undertaker post Royal Assent;
  - SE is not receptive at present to a block shift of either promotion or delivery responsibility for EARL, SAK, Borders and Airdrie-Bathgate to tie or TIS, as envisaged under the tie TIS proposal;
  - SE wishes to settle upon a Promoter selection for EARL as soon as possible;
  - Beyond stating what is not acceptable regarding Promoter identity options, SE has made it clear that they are unable to share the views of their legal and procurement advisers with tie.
- 2. There are a number of constraints which we understand the Scottish Executive perceives. These are:

#### Legally Privileged

### Not for Distribution to or Reliance by any Party other than tie Limited

- 2.1 SE cannot have a stake<sup>1</sup> in any new entity created to promote EARL. This is because the SE considers that ownership would result in any bill being taken as a hybrid bill, needing new (but as yet unwritten) parliamentary process. This will not be sanctioned by Ministers for policy reasons.<sup>2</sup>
- 2.2 SE is concerned that a promoter entity owned by someone other than SE could be instructed to stop work on EARL without the Scottish Executive having any say in the matter.
- 2.3 SE is concerned to achieve clear transfer of risk as to responsibility for EARL promotion and implementation.

#### **Preliminary Recommendation**

Before tie Board takes any final decision, tie obtains a written statement from the Scottish Executive as to how it views the issues raised by the Scottish Executive's requirements for tie's function as Promoter. This will enable a measured review of SE's official position. We do not consider it is correct for the tie Board to assume responsibility for promotion of EARL and management of grant funding without a description from SE of what its immediate and longer term expectations as to the EARL promotion are. Without this, we do not see that (a) tie can advise its sole shareholder with confidence on commercial issues (b) tie will not receive legal advice based on best information and clear guidance from the funder of EARL (c) there is an appreciable risk that solutions are devised which do not address all known facts, intentions and exposure.

#### The Preferred Option for EARL Promoter

- Based upon known requirements, the following options for the identity of the formal Promoter of the EARL Bill exist:
  - CEC
  - tie (in its current form)

<sup>&</sup>lt;sup>1</sup> This appears to cover de minimis or non-controlling interest

<sup>&</sup>lt;sup>2</sup> It should be noted that the advice of parliamentary counsel to tie is that this process is achievable at risk of some cost and delay; this course of action would not, however, suit current EARL timetable requiring bill submission in early May.

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

- Network Rail
- a new single purpose entity (public or private company)
- a non-departmental public body.

Of these five options, the Scottish Executive has (a) indicated informally that using **tie** within the *status quo* is its preferred option (b) rejected the case for a non-departmental public body (NDPB) as promoter. This option<sup>3</sup> would answer many of the legal and practical difficulties which attach to the other four options but we are instructed not to explore the NDPB route as a platform for EARL bill promotion and project implementation on account of SE policy reasons. The table below indicates the key risks and benefits for each of the remaining four options:

Promoter	Benefits	Risks
CEC as Promoter	<ul> <li>no issue in relation to a hybrid bill requirement</li> <li>clear public accountability</li> <li>limited action necessary to formalise mandate</li> </ul>	<ul> <li>lapse of political will</li> <li>not resourced to promote and no natural sponsor within CEC</li> <li>legal authority to promote heavy rail bill requires CEC full Council Section 82 resolution to mitigate vires risk</li> <li>decision-making, communication and scope of authority between CEC, tie and SE cumbersome and ill defined</li> <li>tie ends up remaining "de facto" promoter without direct authority</li> </ul>
NR as Promoter	<ul> <li>no issue as to hybrid bill</li> <li>some English experience as promoter</li> <li>logical due to position as ultimate asset owner and operator</li> <li>EARL commissioning and project acceptance facilitated</li> </ul>	<ul> <li>likely to cause lengthy delay</li> <li>tie role uncertain or non-existent</li> <li>serious doubt over NR corporate delivery capacity</li> <li>negative impact on project credibility in market</li> <li>BAA unlikely to accept</li> <li>departure from stated SE stance in</li> </ul>

<sup>3</sup> We understand this rejection is for policy reasons.

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

Promoter	Benefits	Risks
tie as Promoter	<ul> <li>continuity and private bill process</li> <li>expertise in place with planned resource</li> <li>enhancement of tie viability as implementer on other rail projects in Scotland</li> <li>decision-making simplified and control of information, negotiations and process under one roof</li> <li>tie's authority as a counter party for private sector reinforced</li> <li>preserves EARL programme and momentum</li> <li>communication channels established</li> </ul>	<ul> <li>EARL consultation</li> <li>interface with SE remains fragile and SE ability to oversee NR limited</li> <li>EARL unsuitable in size and profile as a NR pilot project</li> <li>EARL relegated by other priorities</li> <li>potentially stricter proof as to tie's ability to deliver post Royal Assent since tie is subsidiary of local authority</li> <li>direct responsibility for costs of promotion and liabilities incurred in obtaining and preparing for use of powers (ie CPO compensation, discharge of third party obligations)</li> <li>need for CEC position on support for EARL to be formalised clearly</li> <li>no commitment for SE to tie's role post Royal Assent</li> <li>condition attached to grant funding may create (a) hybrid bill or (b) procurement issue on any transfer of authorised undertaker role as instructed by SE</li> <li>possible VAT consequences for CEC</li> </ul>
EARL Promoter Entity (EPE I) (private)	<ul> <li>Bill can remain Private Bill since EPE entity unconnected to SE</li> <li>dedicated project delivery vehicle created for EARL</li> <li>tie can direct and influence establishment of EPE and supply of EPE resources</li> <li>removal of CEC link to national project</li> <li>assures SE that change of priority by CEC could not influence EPE capacity to service EARL</li> </ul>	<ul> <li>EPE not subject to public accountability regime</li> <li>potential procurement and increased state aid risk issues with selection of private sector party for promoter role and grant funding receipt</li> <li>EPE unusual status for party promoting £500m public infrastructure project</li> <li>conditions imposed on grant funding from SE may create (a) hybrid bill (b) procurement issue on transfer of authorised undertaker role as instructed by SE</li> </ul>

#### DRAFT

### Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

Promoter	Benefits	Risks				
EARL Promoter Entity (EPE II) (public)	<ul> <li>no hybrid bill issue provided guarantor is not SE</li> <li>removal of CEC link to national project, assuring guarantor is not CEC</li> <li>EPE subject to public accountability regime (ie the procurement/regime, project preparation guidelines, freedom of information obligations)</li> <li>as for EPE I</li> <li>less risk of procurement issue selecting EPE II as promoter</li> </ul>	<ul> <li>conditions attaching to grant funding create (a) hybrid bill (b) procurement issue on delegation of authorised undertaker role as instructed by SE after Royal Assent</li> <li>no obvious public sector owner for EPE II</li> <li>time required to authorise</li> </ul>				

### tie as formal Promoter of EARL

4. We are instructed to develop this option in terms of (a) the actions/issues requiring legal input and (b) analysis of assessable risk borne by tie in its capacity of Promoter. There are several issues we have advised previously are necessary in order to make the transition to tie becoming formal Promoter. Each of these is tabled below, alongside the action currently recommended:

	ISSUE	ACTION
1.	With tie as Promoter of EARL, CEC is not	CEC to confirm that it shares this view and to
	actively involved in promotion of EARL so	inform tie accordingly in writing.
	that no resolution pursuant to Section 82 of	
	the Local Government (Scotland) Act 1973	
	is required	

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

	ISSUE	ACTION
2.	Amendment to <b>tie's</b> Operating Agreement with CEC. Advices of June and August 2004 stated how the agreement requires revision to safeguard <b>tie's</b> activity on EARL being within the contractual mandate. We understand <b>tie</b> is revising the agreement to reflect cessation of activity on congestion charging.	Revisions to the Operating Agreement to be agreed between <b>tie</b> and CEC and approved by <b>tie</b> Board and by CEC (by full Council Resolution) <sup>4</sup> . <b>tie</b> Board to resolve that assumption of EARL Promoter role is in best interests of <b>tie</b> and that the Operating Agreement should be amended.
3.	Provision to underwrite <b>tie's</b> liabilities as Promoter of EARL. Since <b>tie</b> is required to indemnify CEC in respect of liabilities and to third parties, this will be of interest to CEC, unless the Operating Agreement is amended to remove this obligation.	Document (with CEC and SE) how tie is to undertake EARL Promoter responsibilities without recourse to CEC budget, in order that tie Board can be satisfied that tie will be in a position to discharge financial obligations incurred for the promotion of EARL. This will entail discussion with SE as to the precise terms of continued grant funding.

### 5. RISKS

The assumption of the EARL Promoter's role carries exposure to risk. The following represent those legal and commercial risks we have identified. We have previously highlighted to **tie** project management adverse impact on **tie** project delivery capability.

	Risk	Mitigation
1.	Scottish Executive asserts influence during	Continual information flow between tie and
	promotion of EARL bill which is not	SE. Clear decision making process at tie

<sup>4</sup> We consider that tie's authority to promote EARL should be supported by a Council resolution so that due diligence on this matter can be answered in the future.

## DRAFT

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

	Risk	Mitigation
	consistent with <b>tie's</b> position and <b>tie's</b> credibility damaged.	Board level. Reliance on governance structure and properly documented stakeholder commitment.
2.	Scottish Executive wishes to retain full discretion to instruct <b>tie</b> to transfer authorised undertaker authority post Royal Assent, creating exposure to bill being challenged as hybrid (at Final Stage) or appointment of authorised undertaker being subject to procurement regulations.	Advise Scottish Executive as to potential risks of retaining discretion over the grant of powers under the bill. Risk of procurement challenge sits with Scottish Executive.
		Ensure <b>tie's</b> ability to fund defence against either legal objection to bill or challenge regarding procurement is underwritten by the Scottish Executive. Budget for (time and cost) and factor into promotion workable arrangements which permit <b>tie</b> to step out of contractual commitments continuing post Royal Assent with minimum commercial exposure. Advise Scottish Executive of potential effect on overall deliverability of EARL.
3.	tie enters commitment with third party in order to promote EARL but Scottish Executive unable to make decision to support the commitment or delays the decision resulting in increased cost or unplanned liability.	Consult with Scottish Executive and develop decision-making boundaries recognising this creates "captive" Promoter risk.
4.	Since <b>tie</b> has no long term interest, <b>tie</b> project implementation strategy not fully developed	tie to be suitably protected from incurring residual liabilities arising from the requirement

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

	Risk	Mitigation
	and Parliament requires close scrutiny of implementation and funding of delivery.	for change of Authorised Undertaker.
5.	tie responsible for primary investigation regarding availability of SE funding for EARL and to Article 87 of the Treaty of Rome (State Aid).	tie to obtain comprehensive advice on the issue and clearance from SE to take further action as necessary, including referral to UK authorities.
6.	The formal role as EARL Promoter highlights the direct interaction between EARL and Edinburgh Tram Line 2 and <b>tie's</b> responsibility for dovetailing the economic cases.	tie prepares its evidence for the further stages of Tram Line 2 in close co-ordination with EARL.
		The issue is discussed in detail by CEC and SE so that <b>tie</b> is not caught representing conflicting aims.
		tie's mandate from CEC reconciles tie's role on both projects.
7.	tie as Promoter impacts invoicing arrangements with CEC, so as to create irrecoverable VAT risk.	We understand that this issue has been considered by CEC following initial advice from DLA Piper/John Kennedy & Co in June last year. We have no further information on this matter.
8.	tie at arm's length from CEC planning and roads functions.	Clear agreement on how EARL (through tie) continues to have access to and support from CEC at working level.
9.	Procurement challenge by third party or spontaneous objection by EU Commission stimulated either by assumption of Promoter role or the change of authorised undertaker.	Understanding of the legal risk and, so far as possible, structuring of arrangements to minimise eventuality. Acceptance by SE that the risk is theirs.

### DRAFT

## Legally Privileged Not for Distribution to or Reliance by any Party other than tie Limited

	Risk	Mitigation
10.	Claims are made against <b>tie</b> with regard to compensation payments by parties affected by EARL.	tie authorised to settle claims made and is indemnified by any authorised undertaker.
11.	Project curtailed for financial or political reasons.	tie to be held harmless by SE through grant funding to meet any liabilities.
12.	Change of CEC administration affects tie's ability to act as effective Promoter.	Ability for tie to resign as Promoter

DLA Piper 16 March 2005

AF/LED/310299/7/4500914



Agenda Item 6c

## **Heavy Rail**

c)SAK - Project and Financial Progress Report



Paper to:	tie Board 21 <sup>st</sup> March 2005
Subject:	Commercial & in Confidence Stirling-Alloa-Kincardine
Subject.	(Project Manger - Richard Hudson)
From:	Paul Prescott
Date:	16 <sup>th</sup> March 2005

### **Project Governance**

The suite of contracts between **tie**, Clackmannanshire Council and Jacobs Babtie has now been signed.

#### Asset Protection Agreement

A revised draft of the APA was sent to NR on the 11<sup>th</sup> March 2005 reflecting the output from our meeting with them on the 1<sup>st</sup> March 2005 and subsequent discussions between NR and the Scottish Executive. Some technical matters remain to be resolved, plus the significant issue of specific implement.

We are targeting agreement on all of these by the end of this month, so that the APA can go to NR's Third Party Enhancement Panel in London on the 4<sup>th</sup> May 2005.

#### **Project Cost**

There have been a number of useful meetings with Network Rail's engineers, and much progress in the understanding of the assumptions and derogations contained within the project scope and their implications on project risk. In addition, **tie** have separately been negotiating with FNJV regarding their management fees.

The output from both these initiatives will materialise in the form of a revised target cost which is expected by 23<sup>rd</sup> March 2005.

The agents of Diageo have been instructed to commence the preliminary design and planning applications for their proposed new site at Bonnybridge. This will enable more meaningful discussions with them regarding the level of compensation in connection with their Carsbridge site. This is still currently estimated at £3.6m but it expected that this will be reduced to circa £2m.

The issue of shallow mineworkings remediation has been further developed and a strategy agreed. This involves further investigation to provide the NR Mining Engineer with sufficient data to evaluate the project team's proposals. The likely solution will involve a combination of speed limit reduction on the freight section of the route and some form of consequential risk management, plus a degree of discrete grouting.

#### Site Progress

Devegetation works is in progress. We plan to have all the trees on the route felled by` the 18<sup>th</sup> March and all the stumps either removed or ground down by the end of the month. Public reaction has been very positive now that there is at last physical evidence of activity.

#### Programme

Subject to satisfactory resolution of the above issues, we expect to have Scottish Executive approval by the end of March 2005, Council Approval during April and Network Rail Approval on the 4<sup>th</sup> May. This will allow the contract for Phase 2 to be awarded early in May.

RH/PGP 16.3.05

Report for Month Ending:	28-Feb-05		Pro	ject Manager:	Richard H	udson
Start Date:					30-Apr-06	
<b>Overall Progress Status</b>		Expendit	ure 2004/5	2	-	Life Funding
		Expendit			Troject	
			1.00%			100
Progress Key:					Finance Key	
In track for successful completion as					Within 10% o	the second s
ssues have arisen which may delay con ssues have arisen which will delay con		discussion/direct	10n.		10 - 20% out >20% outside	
ssues have ansen which will delay con	npietion.	Original Start	Original	Revised	Progress	Progress State
Critical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
. Parliamentary Approval		1-Jul-04	1-Jul-04	Completion	(115,11,C) C	(0,1,1)
. Royal Assent		10-Aug-04	10-Aug-04		C	
. Submit Commissioning Report		31-Jul-04	31-Jul-04		C	
Appoint GI Contractor		23-Jul-04	23-Jul-04		С	
Agree Asset Protection Agreement w	vith NR	27-Aug-04	27-Aug-04	4-Apr-05	IP	CHERRY DOLL
Agree Target Cost and Programme	04 m	25-Oct-04	25-Oct-04	23-Mar-05	IP	
Asset Protection Agreement Signed	by NR	10-Dec-04	10-Dec-04	1-May-05	NS	
. Completion - Phase 1		10-Dec-04	10-Dec-04	1-May-05	IP	No. Charling 1
. Commencement - Phase 2	and an and a second	3-Jan-05	30-Apr-06	1-May-05	NS	
0. Line Opening			30-Apr-06	30-Dec-06	NS	
			Original Cost	Start of Year	Current	
	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
revious Years	£0	£0	£0	£0		£0
004/5	£164,937	£164,937	£164,937	£164,937	£164,937	
005/6	£0	£0	£0	£0		£0
006/7	£0	£0	£0	£0	£0	£0
uture Years	£0	£0	£0	£0		£0
otal for Project Life Cycle	£164,937	£164,937	£164,937	£164,937	£164,937	£0
E180,000 £160,000 £140,000 £120,000 £00,000 £80,000 £40,000 £0						Cost (Cum) ————————————————————————————————————
	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	1		Cost (Cum) — Current Year
£160,000 £140,000 £120,000 £80,000 £80,000 £40,000 £20,000 £0			Nov-04 Dec-04	1		(Cum) <u>−</u> Current Year Budget
£160,000 £140,000 £100,000 £80,000 £80,000 £40,000 £0 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04	Nov-04 Dec-04	1	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £80,000 £80,000 £40,000 £0 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04 Project Lif	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £80,000 £80,000 £00 £00 £00 £00 £00 £00 £00 £00 £0		Sep-04 Oct-04 Project Lif	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £80,000 £40,000 £00 £0 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04 Project Lif	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £80,000 £40,000 £0 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04 Project Lif	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £40,000 £0 £0,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) Current Year Budget (Cum)
£160,000 £140,000 £100,000 £40,000 £0 £0,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
£160,000 £140,000 £100,000 £40,000 £0 £0,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum)
£160,000 £140,000 £100,000 £00,000 £0,000	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) Current Year Budget (Cum)
E160,000 E140,000 E120,000 E00,000	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Sep-04  Project Lif  Project Lif  Action:	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
E160,000 E140,000 E120,000 E00,000 E00,000 E00 E00 E00 E00 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Sep-04  Project Lif  Sep-04  Project Lif  Contemporation  Project Lif  Project Lif  Sep-04  Project Lif  Project Lif  Project Lif  Sep-04  Project Lif  Project Lif Project	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	Cost (Cum) 
E160,000 E140,000 E120,000 E00,000 E00,000 E00,000 E00 Apr-04 May-04 Jun-04 80,000 E00 Apr-04 May-04 Jun-04 80,000 E00 Apr-04 May-04 Jun-04 80,000 E00 Apr-04 May-04 Jun-04 B0,000 E00 E00 Apr-04 May-04 Jun-04 B0,000 E00 E00 Apr-04 May-04 Jun-04 B0,000 E00 E00 E00 Apr-04 May-04 Jun-04 B0,000 E00 E00 E00 E00 E00 Apr-04 May-04 Jun-04 B0,000 E00 E00 E00 E00 E00 E00 E00	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Project Lif  Solution  Project Lif  Control  Project Lif  Control  Project Lif  Project Lif  Project Lif  Project Lif  Project Lif  Project Cost Project Project Cost Project Project Cost Project Project Cost Project	Nov-04 Dec-04	Jan-05 Feb-0	All commenced tailed review b s and the minex	Cost (Cum) — Curren Year Budget (Cum)
E160,000 E140,000 E120,000 E00	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Project Lif  Control of the APA with Nee  Project Cost	Nov-04     Dec-04       e	Jan-05 Feb-0	All Andrews by and the miner and the miner en set up to c ve for presenta	Cost (Cum) — Current Year Budget (Cum)
E160,000 E140,000 E120,000 E00	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Source of the APA with Ne beginning of April	Nov-04 Dec-04	Jan-05 Feb-0	All Andrews by the second seco	Cost (Cum) — Curren Year Budget (Cum)
E160,000 E140,000 E120,000 E00,000 E00,000 E00,000 E00 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Solution  Solution  Project Lif  Project Lif  Project Lif  Solution  Project So	Nov-04 Dec-04	Jan-05 Feb-0	All	Cost (Cum) — Curren Year Budget (Cum)
E160,000 E140,000 E120,000 E00	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Solution  Solution  Project Lif  Project Lif  Project Lif  Solution  Project So	Nov-04 Dec-04	Jan-05 Feb-0	All	Cost (Cum) — Curren Year Budget (Cum)
E160,000 E140,000 E120,000 E00	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  South of the set of the s	Nov-04 Dec-04	Jan-05 Feb-0	All	Cost (Cum) Curren Year Budgel (Cum) Solution Contention Solution Solution Contention Solution Contention Solution Solution Solution Contention Contention Solution Solution Solution Solution Contention Solution So

## TRS00018615\_0106



Agenda Item 7

## Tram

- a) Project Progress Reports\*
- b) Procurement \*
- c) OBC Status
- d) Parliamentary Process \*



Agenda Item 7a

# Tram

## a) Project Progress Reports\*

Transfer for screening looppelion as programmed. Transfer for screening looppelion and		28-Feb-05		Pro	ject Manager:	Kevin Mur	rray
Overall Progress Status       Expenditure 2004/5         Togress Key:       Intro: K or successful completion as programmed.       Finance Key:         Writin 10% of estimate intro: Successful completion or regired discussion/direction.       Dis 20% costide estimate 20% costide estidex0% costide estimate 20% costide estidex0% co	Overall Progress Status				End Date:		
Progress Key mark. for accessful completion as programmed. Such Aver arises which may delay completion or require discussion/direction. Such Aver arises which may delay completion or require discussion/direction. Such Aver arises which may delay completion. Progress Rev arises which may delay completion arises are arrived by a set of the progress Rev arises which may delay completion arises are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arrived by a set of the progress Rev are arear			Expendit	ure 2004/5	Late Dutter	Project	Life Funding
Tragers Ko: Track for successful completion as programmed. Since Mare arisen which may delay completion as programmed. The factor for the full of the formation of the full of the formatio		1. A.	Expendit	ure 2004/5		TTOJECT	Life Funding
ha track for successful completion as programmed. Support Part of Support Parliamentary Documents Previous Years Previous Y			and the second second	146%			
<pre>suse have arisen which my delay completion. ID = 20% outside estimate sizes have arisen which my delay completion. ID = 20% outside estimate part of the part of the part</pre>					and an an and the		
suses have arisen which will delay completion. Periods Version Versio							
Original Date     Original Date     Description     Revised     Progress Sn (Gx, R)			discussion/direc	tion.		and a second second second	
Date       Completion       Completion       Completion       (NS.IP.C)       (G.Y.R)         Support Parliamentary Decoust Lading to Royal Asset       1-Jah-04       24-Dec-05       IP         DPOF Appointment of Operator       5-Jan-04       20-Dec-05       IP         Pablication & Making of TRO's       6-Jan-04       1-Jul-06       IP         Pablication & Start of Year       Correct       Variance       Variance         Trovious Years       64.952.23       64.952.201       64.952.23.12       64.952.23.12         Pablication & Start of Year       Edition Correct       60       Edition Correct       Foresst         Pablication Project Life Cycle       56.925.000       E	ssues have arisen which will delay comp		0.1.1.1.01.4	Orisinal	Derival		
Perpent and Deposit Parlimentary Documents Support Parlimentary Process Leading to Koyal Asset 1 Jano 4 24-Dec-03 DPOF Appointment of Operator 24-Data 24-Dec-05 DPOF Appoint Appoint Appoint Appointment of DPOF 24-Data 24-Dec-05 DPOF Appoint Appoint Appointment of DPOF 24-Data 24-Dec-05 DPOF Appoint Appoint Appoint Appointment of DPOF 24-Data 24-Dec-05 DPOF Appoint A	Cuitical Bath / Milestone Itoms					0	
Support Parlimentary Process Leading to Royal Assert 1-Jan-04 24-Dec-05 IP DPOF Appointment of Operator 3-Jan-04 20-Dec-05 IP Pablication & Making of TRO's 6-Jan-04 1-Jul-06 IP Pablication & Making of TRO's 6-Jan-04 IP Pablication & Making OF Pablication IP Pablication IP Pablication IP Pablication & Making OF Pablication IP Pablication IP Pablicati In Pablication IP Pablication IP Pablication IP Pablicat		cuments			Completion		(G, I, K)
DPOE       2-U-U-3       2-V-U-3       C         DPDE Appointment of Operator       3-Jan-04       1-Jul-06       IP         Publication & Making of TRO's       6-Jan-04       1-Jul-06       IP         Publication & Making of TRO's       6-Jan-04       1-Jul-06       IP         Publication & Making of TRO's       6-Jan-04       1-Jul-06       IP         reviews Years       £4,952,233 <td></td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td></td>						-	
Third Party & Stakeholder Linston       5-Jan-04       20-Dec-05       IP         Problemion & Making of TRO's       6-Jan-04       1-Jul-06       IP         Image: Problemion & Making of TRO's       6-Jan-04       1-Jul-06       IP         Image: Problemion & Making of TRO's       6-Jan-04       1-Jul-06       IP         Image: Problemion & Making of TRO's       Funding       Budget       Estimate       Contract         Product       Funding       Budget       Estimate       Contract       Forceast         Variance       Ed.052,237       Ed.052,237       Ed.052,237       Ed.052,237       Ed.052,237         D08/5       El.072,765       E		<u>B</u> to resput rissouri				1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
Image: Signal provide states       Finding       Original Cost       Start of Year       Current       Variance         reviews Years       £4,952,237       £	. Third Party & Stakeholder Liaison					IP	
Funding       Budget       Estimate       Cost Estimate       Forecast       Variance         trevious Years       £4,952,23	. Publication & Making of TRO's		6-Jan-04	1-Jul-06		IP	
Funding       Budget       Estimate       Cost Estimate       Forecast       Auriance         trevious Years       £4,952,23					-		
Funding       Budget       Estimate       Cost Estimate       Forecast       Auriance         trevious Years       £4,952,23							
Funding       Budget       Estimate       Cost Estimate       Forecast       Variance         004/5       £1,072,763       £1,072,76							
Funding       Budget       Estimate       Cost Estimate       Forecast       Variance         0845       £1,072,765							
Funding         Budget         Estimate         Cost Estimate         Forecast         Variance           004/5         £1,072,763         £1,				Original Cost	Start of Year	Current	
revious Years <u>\$4,952,23</u> £4,952,23 £4,952,25 £4,952,25 £4,952,25 £4,952,25 £4,952,25 £4,952,25	The second s	Funding	Budget		the second second second second		Variance
Delds Delds Dels6 Dels6 Dels6 Dels6 Dels7 Dels7 Dels7 Dels6 Dels7 Dels6 Dels7 Dels7 Dels6 Dels7 Dels7 Dels6 Dels7 De	revious Years	0	-	£4,952,237	£4,952,237	£4,952,237	£0
DBSK6       E0	004/5					£1,562,913	-£490,150
there Years 20 20 20 20 20 20 20 20 20 20 20 20 20	005/6	£0	£0	£0	£0	£0	£0
oral for Project Life Cycle       £6,025,000       £6,025,000       £6,025,000       £6,025,000       £6,025,000       £6,515,150       £490,150         c1 480,000       2004/5       2004/5       Cost       Cos		OF CA.			- Constanting of the second	12127	252 S.7.7
E 1 500.000 E 1 500.000 E 200.000 E 200.0							
Example of the second provides an accurate overview of the project progress and finance." Project Director's signature:	otal for Project Life Cycle	10,025,000	10,025,000	20,025,000	20,025,000	20,515,150	-2490,150
1000000 50000000 50000000 50000000 500000000	Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-(	)5 Mar-05	(Cum)
5,00,00 400,000 5,00	7,000,000		Project Lif	e			
4,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000 1,000,000 2,000,000	6,000,000	++++					
3,00,000 1,000,000,000 1,000,00	5,000,000						
2,00,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000 1,000,000,000,000 1,000,000,000,000 1,000,000,000,000,000 1,000,000,000,000,000 1,000,000,000,000,000,000,000,000,000,0							
1,000,000 20 20 20 20 20 20 20 20 20	3 000 000						
The partiamentary committee took evidence from a range of winesses including the promoter from 3 November 2004 and this concluded on 11 January. It published its Preliminary Stage report on 16 February recommending that the Bill proceed as a Private Bill to the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol.               Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the right from the this report provides an accurate overview of the project progress and finance."			المحتكم وراد			_	
Lifetime Budget (Cum)     Actual/Forecast Cost (Cum)     Actual/Forecast Cost (Cum)     Confirm that this report provides an accurate overview of the project progress and finance."     Project Director's signature:     Project Director's signature:	2,000,000				<del></del>		
Lifetime Budget (Cum)     Actual/Forecast Cost (Cum)     Actual/Forecast Cost (Cum)     Confirm that this report provides an accurate overview of the project Director's signature:     Project Director's signature:     Project Director's signature:     Project Director's signature:	2,000,000 1,000,000 £0	· · · · · · · · · · · · · · · · · · ·					
ummary of Key Points and suggested course of action: The parliamentary committee took evidence from a range of witnesses including the promoter from 3 November 2004 and this concluded on 11 January. It published its Preliminary Stage report on 16 February recommending that the Bill proceed as a Private Bill to the Consideration Stage and that the general principles of the Bill should be agreed to. The recommendations of the report were approved unanimously by the Scottish Parliament on 02 March 2005. No decision on the format or programme has been decided by the Private Bills Unit for the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre.  I confirm that this report provides an accurate overview of the project progress and finance." Project Director's signature: Project Director's signature:	2,000,000 1,000,000 £0	and an an an an	and the day of an	and a a a co co co	00, 00, 00, 00, 00, 00,	00,00,00,00	not not not lears
ummary of Key Points and suggested course of action: The parliamentary committee took evidence from a range of witnesses including the promoter from 3 November 2004 and this concluded on 11 January. It published its Preliminary Stage report on 16 February recommending that the Bill proceed as a Private Bill to the Consideration Stage and that the general principles of the Bill should be agreed to. The recommendations of the report were approved unanimously by the Scottish Parliament on 02 March 2005. No decision on the format or programme has been decided by the Private Bills Unit for the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre.  I confirm that this report provides an accurate overview of the project progress and finance." Project Director's signature: Project Director's signature:	2,000,000 1,000,000 £0	A Jan Fab Mar An Mar			ALOS NON JOURD JILOS DO GO	00°+00 00 00 38	and soon ward ward
The parliamentary committee took evidence from a range of witnesses including the promoter from 3 November 2004 and this concluded on 11 January. It published its Preliminary Stage report on 16 February recommending that the Bill proceed as a Private Bill to the Consideration Stage and that the general principles of the Bill should be agreed to. The recommendations of the report were approved unanimously by the Scottish Parliament on 02 March 2005. No decision on the format or programme has been decided by the Private Bills Unit for the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre. It confirm that this report provides an accurate overview of the project progress and finance." Project Director's signature: Project Director's signature:	2,000,000 1,000,000 £0	-	Lifetime Budget	t (Cum)	21.05 100 100 100 100 000 000	00 00 40 00 00 3	FULLE REAL
published its Preliminary Stage report on 16 February recommending that the Bill proceed as a Private Bill to the Consideration Stage and that the general principles of the Bill should be agreed to. The recommendations of the report were approved unanimously by the Scottish Parliament on 02 March 2005. No decision on the format or programme has been decided by the Private Bills Unit for the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre.  I confirm that this report provides an accurate overview of the project progress and finance." Project Director's signature:	2,000,000 1,000,000 £0 Provides parts that the the share with the share of the shar		Lifetime Budget     Actual/Forecast	t (Cum)	And Wat you and Mark Back Ca	00 000 400 000 30	PUT POT WALL TOUR
principles of the Bill should be agreed to. The recommendations of the report were approved unanimously by the Scottish Parliament on 02 March 2005. No decision on the format or programme has been decided by the Private Bills Unit for the Consideration Stage of the parliamentary process, which will follow. Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre.	2,000,000 1,000,000 <u>E0</u> Provides the part to the set of the se		Lifetime Budget     Actual/Forecast	t (Cum)	and the right read of the read	00 00 400 000 30	FUND FOR THE FUND
Negotiations are ongoing with objectors in general accordance with the Phasing protocol. Tram Line One costing for 2004/05 includes an element of cross funding from Tram Line Two, which reflects work carried out on the common section and the significant issues requiring resolution in the city centre. I confirm that this report provides an accurate overview of the project progress and finance." Project Manager's signature: Project Director's signature:	presides weeks of work work with the set of	ested course of om a range of witnes	<ul> <li>Lifetime Budget</li> <li>Actual/Forecast</li> <li>action:</li> <li>ses including the provided the</li></ul>	t (Cum) t Cost (Cum) romoter from 3 Nover	nber 2004 and this ca	oncluded on 11.	January. It
I confirm that this report provides an accurate overview of the project progress and finance." roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 protection of the Bill should be agreed to. The	ested course of om a range of witnes ebruary recommendin recommendations of	<ul> <li>Lifetime Budget</li> <li>Actual/Forecast</li> <li>action:</li> <li>ses including the ping that the Bill proof the report were set</li> </ul>	t (Cum) t Cost (Cum) romoter from 3 Nover seed as a Private Bill t approved unanimou	nber 2004 and this ca o the Consideration 1 sly by the Scottish F	oncluded on 11. Stage and that th arliament on 0	January. It e general 2 March 2005.
I confirm that this report provides an accurate overview of the project progress and finance." roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes abruary recommendai recommendations of een decided by the P	<ul> <li>Lifetime Budget</li> <li>Actual/Forecast</li> <li>action:</li> <li>ses including the ping that the Bill proof the report were erivate Bills Unit foil</li> </ul>	t (Cum) t Cost (Cum) romoter from 3 Nover teed as a Private Bill approved unanimou the Consideration St	nber 2004 and this ca o the Consideration 1 sly by the Scottish F	oncluded on 11. Stage and that th arliament on 0	January. It e general 2 March 2005.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 £0 provides part and sugged provides and sugged the parliamentary committee took evidence for published its Preliminary Stage report on 16 Fe principles of the Bill should be agreed to. The No decision on the format or programme has b Negotiations are ongoing with objectors in gen	ested course of om a range of witnes ebruary recommendations o een decided by the P eral accordance with	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoc	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It e general 2 March 2005. iich will follow.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. iich will follow.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. tich will follow.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. tich will follow.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. tich will follow.
roject Manager's signature: Project Director's signature:	2,000,000 1,000,000 E0 Presides and the start of the s	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. tich will follow.
	2,000,000 1,000,000 50 50 50 50 50 50 50 50 50	ested course of om a range of witnes ebruary recommendations of een decided by the P terral accordance with n element of cross fu	Actual/Forecass action: ses including the ping that the Bill proof of the report were a rivate Bills Unit for the Phasing protoco	t (Cum) t Cost (Cum) romoter from 3 Nover ceed as a Private Bill t approved unanimou r the Consideration St iol.	mber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliamen	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It te general 2 March 2005. tich will follow.
Date: 08/03/2005 Date: 10 3 5 08/03/20	2,000,000 1,000,000 E0 Provide State of the State of t	ested course of om a range of witness ebruary recommendations of eeen decided by the P terral accordance with n element of cross fur ty centre.	Lifetime Budget Actual/Forecast action: ses including the ping that the Bill proo of the report were a rivate Bills Unit for the Phasing protoc	t (Cum) t Cost (Cum) romoter from 3 Nover seed as a Private Bill t approved unanimou the Consideration St sol. ine Two, which reflect	nber 2004 and this co o the Consideration 3 sly by the Scottish F age of the parliament ats work carried out o	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It e general 2 March 2005. iich will follow.
Date: [0]\$]5	2,000,000 1,000,000 50 provide protection of the Bill should be agreed to. The No decision on the format or programme has b Negotiations are ongoing with objectors in gen Tram Line One costing for 2004/05 includes an significant issues requiring resolution in the cit	ested course of om a range of witness ebruary recommendations of eeen decided by the P terral accordance with n element of cross fur ty centre.	Lifetime Budget Actual/Forecasi action: ses including the ping that the Bill proc of the report were a rivate Bills Unit for the Phasing protoc nding from Tram L	t (Cum) t Cost (Cum) comoter from 3 Nover seed as a Private Bill t approved unanimou r the Consideration St tol. ine Two, which reflect t progress and fin	nber 2004 and this or o the Consideration 3 sly by the Scottish F age of the parliamen ets work carried out of ance. <sup>29</sup>	oncluded on 11. Stage and that th arliament on 0 tary process, wh	January. It e general 2 March 2005. iich will follow.
	2,000,000 1,000,000 50 Provide and the start of the st	ested course of om a range of witness ebruary recommendations of eeen decided by the P teral accordance with n element of cross fur ty centre.	- Lifetime Budget - Actual/Forecast action: ses including the progent were environment if the report were environment inding from Tram L we of the project	t (Cum) t Cost (Cum) comoter from 3 Nover seed as a Private Bill t approved unanimou r the Consideration St tol. ine Two, which reflect t progress and fin	nber 2004 and this co o the Consideration is sly by the Scottish F age of the parliamen ats work carried out of ance." S signature:	oncluded on 11. Stage and that th <b>'arliament on 0</b> tary process, wh	January. It he general 2 March 2005. hich will follow.
	2,000,000 1,000,000 E0 Presides and a start of the part of the start of the part of the start	ested course of om a range of witness ebruary recommendations of eeen decided by the P teral accordance with n element of cross fur ty centre.	- Lifetime Budget - Actual/Forecast action: ses including the progent were environment if the report were environment inding from Tram L we of the project	t (Cum) t Cost (Cum) comoter from 3 Nover seed as a Private Bill t approved unanimou r the Consideration St tol. ine Two, which reflect t progress and fin	nber 2004 and this co o the Consideration is sly by the Scottish F age of the parliamen ats work carried out of ance." S signature:	oncluded on 11. Stage and that th <b>'arliament on 0</b> tary process, wh	January. It he general 2 March 2005. hich will follow. section and the

Report for Month Ending: 2         Start Date:         Overall Progress Status         Progress Key:         On track for successful completion as prodised arisen which may delay complete arisen which will delay complete arisen and Deposit Parliamentary Doce         Critical Path / Milestone Items         1. Prepare and Deposit Parliamentary Doce         2. Support Parliamentary Process Leading         3. DPOF Appointment of Operator         4. Third Party & Stakeholder Liaison         5. Publication & Making of TROs         Previous Years         2004/5	ogrammed. letion or require di letion.		ure 2004/5 68%			Life Funding 113% of estimate side estimate
Overall Progress Status Progress Key: On track for successful completion as pro Issues have arisen which may delay completes the arisen which will delay complete the arisen which will delay complete the arisen which will delay complete the arisen and Deposit Parliamentary Doce 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	letion or require di letion.	scussion/directio Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	on. Original Completion 24-Dec-03	Revised	<b>Finance Key:</b> Within 10% o 10 – 20% outs >20% outside	1139 : of estimate side estimate
Progress Key: On track for successful completion as pro- Issues have arisen which may delay completion Issues have arisen which will delay complete Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	letion or require di letion.	scussion/directio Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	on. Original Completion 24-Dec-03	Revised	<b>Finance Key:</b> Within 10% o 10 – 20% outs >20% outside	1139 : of estimate side estimate
On track for successful completion as pro Issues have arisen which may delay comp Issues have arisen which will delay complete Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	letion or require di letion.	Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	on. Original Completion 24-Dec-03	Revised	Within 10% o 10 – 20% outs >20% outside	: of estimate side estimate
On track for successful completion as pro Issues have arisen which may delay comp Issues have arisen which will delay complete Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	letion or require di letion.	Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	Original Completion 24-Dec-03	Revised	Within 10% o 10 – 20% outs >20% outside	of estimate side estimate
Issues have arisen which may delay comp Issues have arisen which will delay compl Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	letion or require di letion.	Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	Original Completion 24-Dec-03	Revised	10 – 20% outs >20% outside	side estimate
Issues have arisen which will delay compl Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	etion.	Original Start Date 4-Oct-02 1-Jan-04 2-Jul-03	Original Completion 24-Dec-03	Revised	>20% outside	
Critical Path / Milestone Items 1. Prepare and Deposit Parliamentary Doc 2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years	cuments	Date           4-Oct-02           1-Jan-04           2-Jul-03	Completion 24-Dec-03	Revised		
Previous Years     Previous Years	cuments	Date           4-Oct-02           1-Jan-04           2-Jul-03	Completion 24-Dec-03		Trogress	Progress Statu
Previous Years     Previous Years		4-Oct-02 1-Jan-04 2-Jul-03	24-Dec-03	compretion	(NS,IP,C)	(G,Y,R)
2. Support Parliamentary Process Leading 3. DPOF Appointment of Operator 4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years		2-Jul-03	20-Dec-05	start as shifty	C	(0,1,1)
4. Third Party & Stakeholder Liaison 5. Publication & Making of TROs Previous Years			20 200 05		IP	
5. Publication & Making of TROs Previous Years		5-Jan-04	29-Apr-04		С	
Previous Years			20-Dec-05		IP	
		6-Jan-04	1-Jul-06		IP	
			Original Cost	Start of Year	Current	
	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
2004/5	£2,940,316	£2,940,316	£2,940,316	£2,940,316		
2005/6	£1,838,360	£1,838,360	£1,838,360	£1,838,360	£1,253,368 £221,324	
2005/6	£221,324 £0	£221,324 £0	£221,324 £0	£221,324 £0		£0 £0
Future Years	£0	£0	£0	£0		£0
Total for Project Life Cycle	£5,000,000	£5,000,000	£5,000,000	£5,000,000	£4,415,008	
and the second	Constraint of States	2004/5	and the second second			
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04 Project Life		Jan-05 Feb-0	05 Mar-05	Curren Year Budget (Cum)
25,000,000						
£4,000,000					<u></u>	
23,000,000						
2,000,000						
c1,000,000			1 2			
Playor.		Lifetime Budget ( Actual/Forecast C	Cum)	ACR WAY JUL OF JUL DA CR	60 000 400 000 100	Filling Loads
Summary of Key Points and sugg	ested course of	action:		in a state		
he committee published its Preliminary Stage meneral principles of the Bill should be agreed to ecision on the format or programme has been deteneral accordance with the Phasing protocol. M have submitted a claim for £175k for add precast). the has not accepted this and are contin	. The recommendati ecided for the Consid	ions of the report leration Stage of the d in meeting the p	were approved unan e parliamentary proces	imously by parliam ss. Negotiations are	ent on 23 Febru ongoing with ob	uary. No ojectors in
I confirm that this report provides an a	icourate overview	of the project p	orogress and finance	ce."		
communications report provides an a		F	Project Director'	e signatures		
roject Manager's signature:			injust Director	s signature:		
		09/03/2005	Tojeet Director	s signature: Date:	10/3/5	09/03/20

	28-Feb-05		Pro	ject Manager:	Ian Kenda	Ш
Start Date:				End Date:		
<b>Overall Progress Status</b>		Expendit	ure 2004/5		Project	Life Funding
A stephen and the state			and the second	1.1.1.1.1.1.1		10
rogress Key:			51%	den de la	Finance Key	
In track for successful completion as	programmed				Within 10% of	
ssues have arisen which may delay con		discussion/direct	ion.		10 - 20% out	
ssues have arisen which will delay con					>20% outside	
		Original Start	Original	Revised	Progress	Progress Sta
Critical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
. Prepare and Deposit Parliamentary D		1-Jul-02	23-Dec-03		С	
. Support Parliamentary Process Lead	ing to Royal Asser		24-Dec-05		IP	
DPOF Appointment of Operator		2-Jul-03	29-Apr-04	14-May-04	IP	
Third Party & Stakeholder Liaison		5-Jan-04	20-Dec-05		IP	
Publication & Making of TRO's Phase C1 Start		6-Jan-04	1-Jul-06 30-Jun-06	1-Jul-06	IP	
Phase C1 Start			30-Jun-06 1-Jan-09	1-Jul-06 1-Mar-09		
. Full System Open	-		31-Oct-09	31-Oct-09		
			a de de la com			
	Freedland	Product	Original Cost Estimate	Start of Year	Current	Varia
the New York	Funding	Budget		Cost Estimate	Forecast	Variance
revious Years	£534,000	£534,000	£534,000	£534,000	£534,000	1-203
004/5 005/6	£5,008,000 £0	£5,008,000 £0	£5,008,000 £0	£5,008,000 £0		£2,433,367 -£2,433,367
006/7	£0	£0	£0 £0	£0 £0		£0
uture Years	£0	£0	£0	£0		£0
otal for Project Life Cycle	£5,542,000	£5,542,000	£5,542,000	£5,542,000	£5,542,000	
£2,000,000 £1,000,000						Year
	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	95 Mar-05	Year Cost
£1,000,000		1	Nov-04 Dec-04	4 Jan-05 Feb-0	5 Mar-05	
£1,000,000 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 5,000,000		Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 4,000,000		Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 4,000,000 3,000,000 1,000,000		Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 4,000,000 2,000,000 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			Year Cost
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 4,000,000 2,000,000 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			Year Cost Estima
E1,000,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 Apr-04 May-04 Jun-04 Apr-04 Apr-04 Apr-04 Apr-04 May-04 Jun-04 Apr-04	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Project Lif  Oct-04  Project Lif  Control of the second of	Nov-04 Dec-04	cicipated, primarily cc rrway as is developm Transdev are makin arly stage insights int	ponditioned by the ent of the outlin g a valuable cor o opportunities a	Year Cost Estim
£1,000,000 £0 Apr-04 May-04 Jun-04 5,000,000 5,000,000 4,000,000 2,000,000 1,000,000 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Oct-04  Project Lif  of  physe get of  of  of  physe get of	Nov-04 Dec-04	icipated, primarily co erway as is developm Transdev are makin arty stage insights int t will in aggregate fal	proditioned by the ent of the outlin g a valuable cor o opportunities a l some £2.4m be	Year Cost Estim

TRS00018615\_0111

and a second	28-Feb-05		iamentary Orde Pro	ject Manager:	Willie Fra	ser
Start Date:				End Date:		
<b>Overall Progress Status</b>		Expendit	ure 2004/5		Project	Life Funding
	a second and		92%	i i i i i i i i i i i i i i i i i i i		101
Progress Key: On track for successful completion as					Finance Key Within 10%	
ssues have arisen which may delay co		discussion/direct	ion			tside estimate
ssues have arisen which may delay co		discussion/direct	1011.	STATISTICS.	>20% outside	
ssues have ansen which will delay con	inpretion.	Original Start	Original	Revised	Progress	Progress Stat
Critical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
. Appoint advisory team		-	1-Jul-03		C	
. Preferred Alignment		19-Dec-03	20-Jan-04	20-Feb-04	С	
. Development of Preliminary Finance	ial Case	1-Sep-04	15-Oct-04	26-Nov-04	IP	
. Public Consultation		24-Mar-04	18-May-04		С	
Scheme appraisal (STAG 2)		1-Dec-03	30-Nov-04	TBC	IP	
. Parliamentary Process to Royal Asse	ent	TBC 5-Jan-04	TBC 15-Nov-04	TBC	NS IP	
. Environmental appraisal . Parliamentary Documents (submission)	on of Bill)	5-Jan-04	17-Dec-04	TBC	NS	
. Farnamentary Documents (submission	on or Billy		17-Dec-04	IBC	NS	
			Original Cost	Start of Year	Current	
Communication of the second	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
revious Years	£790,628	£790,628	£790,628	£790,628		
004/5	£1,983,989	£1,983,989	£1,983,989	£1,983,989		
005/6	£725,383	£725,383	£725,383	£725,383		-£155,089
006/7 Suture Years	£0 £0	0£ 0	£0 £0	£0 £0		£0 £0
Total for Project Life Cycle	£3,500,000	£3,500,000	£3,500,000	£3,500,000	£3,500,000	
£1,500,000						
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	05 Mar-05	(Cum) 
Apr-04 May-04 Jun-04		Sep-04 Oct-04 Project Lif	ie	Jan-05 Feb-0	05 Mar-05	Curren year Budge (Cum)
Apr-04 May-04 Jun-04 4,000,000 3,500,000 2,000,000 2,000,000 1,500,000 £0	Jul-04 Aug-04	Sep-04 Oct-04	South of the south			Curren year Budge
Apr-04 May-04 Jun-04 4,000,000 3,500,000 2,500,000 2,500,000 1,500,000 £0 £0 presides for port, the standard and the standard	Jul-04 Aug-04	Sep-04 Oct-04	South of the south			Curren year Budge (Cum)
Apr-04 May-04 Jun-04 4,000,000 3,500,000 2,500,000 1,500,000 £0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E	Jul-04 Aug-04	Sep-04 Oct-04  Project Lift  Project Lift  Solution  Sol	by Congestions Chargin is due to the delay in s significantly less than require s for TL3.	a, Following the recent ubmitting the Bill to Pa uired. There is developen	referendum, this is referendum, this is rrliament and beca nent funding availa	Curren year Budge (Cum)
Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 4,000,000 3,500,000 2,000,000 1,500,000 1,500,000 1,500,000 1,000,000	Jul-04 Aug-04	Sep-04 Oct-04  Project Life  Sep-04  Project Life  Sep-04  Project Life  Sep-04  Sep-04  Project Life  Sep-04  Sep-04 Sep-04 Sep-04  Sep-04 Sep-04 Sep-04 Sep-	be be be be be be be be be be	a, Following the recent ubmitting the Bill to Pa uired. There is developen ct to conduct a review.	referendum, this is referendum, this is rrliament and beca nent funding availa	Curre year Budge (Cum
Apr-04 May-04 Jun-04 4,000,000 3,500,000 2,500,000 1,500,000 £00,000 £0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Source of the project is a constraint of the project is a constraint of the project. This is stage of the project is a constraint of the project. This is stage of the project is a constraint of the project. The risk is berequired for the Score of the project. The risk is berequired for the score of the project costs an freantly less than require	the risk that the TL3 Prive that the project may be accrued to achieve Royal Asset	ag. Following the recent ubmitting the Bill to Pe uired. There is developed of to enable parliamenta to conduct a review. 1 d to next financial year. ate Bill will not be sub topped or delayed. ent against the ARUP fu financial year, TL3 is p ramme. In addition, expe	referendum, this is rritiament and bece nent funding availa ry documents and taking account of This covers work mitted to Parliam nding statement rojected to have rrience on TL1 and	Curren year Budge (Cum)


Agenda Item 7b

## Tram

b) Procurement \*



1

Agenda Item 7c

## Tram

c) OBC Status



Agenda Item 7d

## Tram

d) Parliamentary Process \*



#### Edinburgh Tram – Statutory Processes - Line 1 & 2 Status Report

Private Bills for Edinburgh Tram Line 1 and 2 are currently under consideration by their respective Parliamentary Committees. Both Committees submitted their Preliminary Stage Reports to the Scottish Parliament and the recommendations to proceed to the Consideration Stage were each approved unanimously.

During the Consideration Stage, the detail of the proposals will be considered. The Committees have now each held a meeting to consider how to manage the Consideration Stage. They have proposed groupings of objectors so that similar issues can be dealt with at the one time. They have written to all objectors notifying them of their proposals and inviting them to a meeting to explain and discuss them. The PBU will meet with **tie** separately to discuss the proposed procedures.

The Committees do not intend to meet again until June. Before that, lists of witnesses and the scope of evidence will be submitted, followed by detailed evidence. The evidence statements will be exchanged between **tie** and the objectors and will be taken as read at the Committees. There will be no leading of evidence at the Committees – it will go straight to cross-examination. The PBU has outlined a very demanding work schedule to get through all the objection hearings, with Committees each having all day sessions once a week during June and recommencing in September.

**tie** is still budgeting on completion of the Consideration Stage and the Final Stage, and achieving Royal Assent to the Bills by the end of December 2005 although this depends entirely on the progress of considering the Bills by the Committees.

Meanwhile, there is considerable work ongoing in seeking to resolve the concerns of formal objectors to the Bills. It is intended to try to achieve resolution with as many as possible to avoid the need to appear before the Committees. There are a number of objections on each line where it is unlikely that agreement will be reached and the issues will have to be examined at the Committee hearings.

For Tram Line 1, 196 objections were lodged through the formal objection period following the lodging of the Bill. The Committee considered 10 late objections and found that 9 demonstrated good reason for lateness and were therefore deemed admissible. 7 objections have since been removed and 198 objections remain to the Bill with the following proportions; approximately

- 34 objections have been raised by commercial entities
- 137 objections have been raised on the Roseburn Corridor
- 27 objections have been raised along the waterfront.

In addition to managing and resolving such objection matters it has become necessary to redevelop and negotiate two s75 Agreements through the key waterfront areas. This is being progressed simultaneously through the resolution

Prepared By: Kevin Murray and Geoff Duke Date: 25/02/2005 Board Mtg 22March05 Item X 1 ETL1+2 Status Report



of the technical issues and would preferably be concluded legally in the same timeframe.

For Tram Line 2, 82 objections were lodged through the formal objection period following the lodging of the Bill. The Committee considered 3 late objections and found that they demonstrated good reason for lateness and were therefore also deemed admissible. 9 objections to the Whole Bill have been rejected by the Committee and therefore 76 objections. These can broadly be classified as:

- 2 environmental/heritage
- 5 transport
- 13 residential
- 56 commercial

In parallel with this, the preparation of evidence and consideration of the case to be presented through the Consideration Stage is under way. Key witnesses have been identified and consultations with the QC are reviewing the strategy to be adopted and developing the case to be made. Supporting work is also being prepared to address the key areas of concern raised by the Committee which are summarised below;

- non-user benefits may have been overstated in economic case.
- Design Manual has no formal status in the approvals process.
- not convinced that spirit of full co-operation between Lothian Buses and those parties involved in the promotion has always existed.
- concerns that the social inclusion benefits will not be fully realised unless the Concessionary Fares Scheme is available to tram users.
- concerns about the consultation exercise which with hindsight could have been improved.
- more detailed information on utility diversion costs to be provided.
- main reservations relate to the scale of economic benefits which have been attributed to non tram users and also the need for further reassurances as to the robustness of the patronage forecasts.

Two route alignment changes outwith the LOD are being considered – one on the common section at Haymarket Yards and one at the Gyle. The purpose of the changes is to reach resolution with objectors. However, by going outside the LODs, a "Bill within a Bill" procedure is triggered that, for the area affected, will involve consultation, notification and an objection period. Technical work is progressing as well as an assessment of the pros and cons of making the amendment.

Detailed consideration of the impact of any of the early design work by the Tram Implementation Team is being undertaken. This includes a detailed review of property impacts and potential compensation implications which will inform the Outline Business Case.

Prepared By: Kevin Murray and Geoff Duke Date: 25/02/2005 Board Mtg 22March05 Item X 1 ETL1+2 Status Report



Agenda Item 8

# **Other Projects**

- a) Project Progress Reports\*
- b) Ingliston Park & Ride Lessons Learnt \*
- c) ITIBS \*



Agenda Item 8a

# **Other Projects**

a) Project Progress Reports\*

Report for Month Ending:	28-Feb-05		Pro	ject Manager:	John Saun	ders
Start Date:				End Date:		
<b>Overall Progress Status</b>		Expendit	ure 2004/5		Project	Life Funding
			94%	10 A.		10.
Progress Key: On track for successful completion as	neogenerad				Finance Key Within 10% of	
Issues have arisen which may delay co		e discussion/direc	ction		10 - 20% out	
Issues have arisen which will delay co		e discussion/direc		A CONTRACTOR OF	>20% outside	
Critical Path / Milestone Items		Original Start Date	Original Completion	Revised Completion	Progress (NS,IP,C)	Progress Star (G,Y,R)
1. Update business Case		1-Feb-03	31-Jan-04		С	Line and the
2. Prepare Draft Charging Order and a		1-Feb-03	15-Sep-03		С	
3. Develop and assemble background	material	24-Mar-03	26-Sep-03		С	
4. Draft Charging order to Council		22-Sep-03	30-Sep-03		С	
5. Publication and objection period CC	)	2-Oct-03	28-Feb-04		С	
6. Negotiation. Public inquiry		3-Oct-03 6-Jan-03	2-Jul-04 11-Nov-04	Mid Ion 2005	C	
7. Referendum preparation 3. Prepare application in Detail		6-Jan-03 15-Aug-03	11-Nov-04 15-Nov-04	Mid Jan 2005 Spring 2005	C IP	
9. Final scheme approval by Council		13-Aug-03 12-Nov-04	15-Dec-04	Spring 2005 Spring 2005	IP	
10. AiD to Scottish Executive		Mid Feb 2005	1-Jun-05	Spring 2005	IP	
11.Procurement system Operator		1-May-03	20-Jul-05		IP	
12. Retail Impact study		21-Jan-04	30-Sep-04	Mid Nov 2004	C	
	Funding	Budget	Original Cost Estimate	Start of Year Cost Estimate	Current Forecast	Variance
Previous Years	£2,851,571	£2,851,571	£2,851,571	£2,851,571	and the second s	
2004/5	£1,131,213	£1,131,213	£1,131,213	£1,131,213		
2005/6	£0	£0	£0	£0		£0
2006/7	£0		£0	£0		£0
Future Years	£0	and the second second	£0	£0		£0
Total for Project Life Cycle	£3,982,784	£3,982,784	£3,982,784	£3,982,784	£3,915,206	£67,578
£800,000 £600,000 £400,000 £200,000						cast Cost (Cum)
	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
£200,000 £0 Apr-04 May-04 Jun-04			Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
£200,000 £0 Apr-04 May-04 Jun-04 £4,500,000		Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
£200,000 £0 Apr-04 May-04 Jun-04 £4,500,000 £3,500,000 £3,500,000		Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
£200,000 £0 Apr-04 May-04 Jun-04 £4,500,000 £3,500,000 £3,500,000 £2,500,000		Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
£200,000 £0 Apr-04 May-04 Jun-04 £4,500,000 £3,500,000 £3,500,000 £2,500,000 £2,500,000 £1,500,000		Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E500,000		Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum)
E200,000 £0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E1,000,000 E1,000,000 E0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			(Cum)
E200,000 £0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E1,000,000 E1,000,000 E0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E1,500,000	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E3,500,000 E1,500,000 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04			(Cum) Current Year Budg (Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lit	Nov-04 Dec-04			(Cum)
E200,000 E0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04 Project Lit	Nov-04 Dec-04			(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lit  Source of the second se	Nov-04 Dec-04	و development of the	Store to the store of the store	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lit  Solution  Solution  Solution  Solution  Project Lit  Solution  Solut	Nov-04 Dec-04	و development of the	Store to the store of the store	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lit  Solution  Solution  Solution  Solution  Project Lit  Solution  Solut	Nov-04 Dec-04	و development of the	Store to the store of the store	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lit  Solution	Nov-04 Dec-04	e development of the council's decision a	Congestion Char nd as such final in	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Solution  Solut	Nov-04 Dec-04	e development of the Council's decision at olicitor and the Council	م م م م م م م م م م م م م م م م م م م	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Solution  Solut	Nov-04 Dec-04	e development of the Council's decision at olicitor and the Council	م م م م م م م م م م م م م م م م م م م	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 Extra production of the seferendum production of the seferendum judicial review. Work had also been continuing	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif Project Lif Project Lif Synch July Synch Strong Synch July Strong Synch July Strong Synch July Strong Synch July Synch Strong Synch July Synch July Strong Synch July Strong Sy	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council g / editing of the Stag	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 E0 E1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif Project Lif Project Lif Synch July Synch Strong Synch July Strong Synch July Strong Synch July Strong Synch July Synch Strong Synch July Synch July Strong Synch July Strong Sy	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council g / editing of the Stag	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif Project Lif Project Lif Synch July Synch Strong Synch July Strong Synch July Strong Synch July Strong Synch July Synch Strong Synch July Synch July Strong Synch July Strong Sy	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council g / editing of the Stag	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 E0 E1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif Project Lif Project Lif Synch July Synch Strong Synch July Strong Synch July Strong Synch July Strong Synch July Synch Strong Synch July Synch July Strong Synch July Strong Sy	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council g / editing of the Stag	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif Project Lif Project Lif Synch July Synch Strong Synch July Strong Synch July Strong Synch July Strong Synch July Synch Strong Synch July Synch July Strong Synch July Strong Sy	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council g / editing of the Stag	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E2,500,000 E1,500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Solution	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council etary saving on the lev	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E1,500,000 E0 E500,000 E0 E0 E0 E1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Source of the project  Project Lif  Project Lif  Project Lif  A control of the project  Project Lif  Project Lif Project Lif  Project Lif  Project Lif Project Lif  Project Lif Projet Lif Project Lif Project	Nov-04 Dec-04	e development of the council's decision at olicitor and the Count g / editing of the Stag etary saving on the lev	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E1,500,000 E0 E500,000 E0 E0 E0 E1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Source of the project  Project Lif  Project Lif  Project Lif  A control of the project  Project Lif  Project Lif Project Lif  Project Lif  Project Lif Project Lif  Project Lif Projet Lif Project Lif Project	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council y editing of the Stag etary saving on the lev	Congestion Char nd as such final i cil's QC to assist e 2 STAG docum	(Cum)
E200,000 E0 Apr-04 May-04 Jun-04 E4,500,000 E3,500,000 E3,500,000 E1,500,000 E1,500,000 E500,000 E500,000 E500,000 E500,000 E0 Following the announcement on 22 February cease. All consultants and advisors who were being prepared to allow finalisation of the acc	Jul-04 Aug-04	Sep-04 Oct-04  Project Lif  Project Lif  Source of the project  Project Lif  Project Lif  Project Lif  A control of the project  Project Lif  Project Lif Project Lif  Project Lif  Project Lif Project Lif  Project Lif Projet Lif Project Lif Project	Nov-04 Dec-04	e development of the council's decision at olicitor and the Council y editing of the Stag etary saving on the lev	Congestion Char nd as such final in cil's QC to assist e 2 STAG docum	(Cum)

Report for month Ending.	28-Feb-05		Pro	ject Manager:	Seamus He	aly
Start Date:				End Date:		
<b>Overall Progress Status</b>		Expendit	ure 2004/5		Project	Life Fundin
				i	,	
			92%			i til i til d
rogress Key:					Finance Key	
in track for successful completion as j		discussion/direc	tion		Within 10% out	
ssues have arisen which may delay completion or require ssues have arisen which will delay completion.		discussion/direction.			10 - 20% outside estimate >20% outside estimate	
		<b>Original Start</b>	Original	Revised	Progress	Progress St
ritical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
Complete Evaluation of System Proc		8-Apr-04	14-May-04		С	
Contract with Agreed System Integra	ators (SI)	14-May-04	14-Jun-04		С	
SIs Team Mobilisation Complete	N 1 11 1	14-Jun-04	5-Jul-04		C	
Macro Designs Complete (Business) Technical Designs Complete	Modelling)	5-Jul-04 6-Aug-04	16-Aug-04 8-Nov-04		C C	
Architecture Designs Complete	-	25-Oct-04	6-Dec-04		C	
Prototypes Design and Build Comple	ete	9-Aug-04	28-Oct-04	General Street	C	
Prototype Tests Complete		28-Oct-04	8-Dec-04	. · · bal- page	С	
Complete Evaluation of Stage 1 Desi		20-Dec-04	21-Jan-05	28-Feb-05	С	
). Finalise Stage 2 Contract Schedules		24-Jan-05	18-Feb-05	15-Mar-05	С	
. Exercise Stage 2 Option with Chose	en SI	21-Feb-05	18-Mar-05		С	
and the second second second			Original Cost	Start of Year	Current	
	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
revious Years	£694,159	£694,159	£694,159	£694,159	£694,159	£0
004/5	£2,048,701	£2,048,701	£2,048,701	£2,048,701	£1,890,431	
005/6	£0	£0	£0	£0		£0
006/7	£0	£0	£0	£0		£0
uture Years otal for Project Life Cycle	£0 £2,742,860	£0 £2,742,860	£0 £2,742,860	£0 £2,742,860	07-15	£0
Sur for the jeer have by the	21,7 12,000					
£1,000,000 £500,000						Curr Yea
£1,000,000	Jul-04 Aug-04	Sep-04 Oct-04			05 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04		Sep-04 Oct-04	Nov-04 Dec-04		05 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04			Nov-04 Dec-04		)5 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000		Sep-04 Oct-04	Nov-04 Dec-04		05 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000		Sep-04 Oct-04	Nov-04 Dec-04		)5 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000		Sep-04 Oct-04	Nov-04 Dec-04		05 Mar-05	Curr Year Budg
£0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000		Sep-04 Oct-04	Nov-04 Dec-04		05 Mar-05	Curr Year Budg
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 1,000,000 £500,000		Sep-04 Oct-04	Nov-04 Dec-04		05 Mar-05	Curr Year Budg (Cur
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	Fe	4 Jan-05 Feb-0		Curr Year Budg (Cun
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,000,000 1,500,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	Fe			Curr Year Budg (Cun
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,000,000 1,500,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	re po po po po po po po po po po	4 Jan-05 Feb-0		Curr Year Budg (Cun
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 £0 £0 50 £0 50 50 £0 50 50 50 50 50 50 50 50 50 5	Jul-04 Aug-04	Sep-04 Oct-04 Project Li	1 Nov-04 Dec-04 <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b></b>	4 Jan-05 Feb-0		(Cun Year Budg (Cun
£1,000,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	1 Nov-04 Dec-04 <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b></b>	4 Jan-05 Feb-0		Curr Year Budg (Cun
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 1,500,000 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0 £0	Jul-04 Aug-04	Sep-04 Oct-04	1 Nov-04 Dec-04 <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b></b>	4 Jan-05 Feb-0		Curr Year Budg (Cur
£1,000,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,000,000 1,000,000 £0 Prove State Stat	Jul-04 Aug-04	Sep-04 Oct-04	1 Nov-04 Dec-04 <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b>fe</b> <b></b>	4 Jan-05 Feb-0		Curr Year Budy (Cur
£1,000,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 500,000 500,000 £0 500,00	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Source States Sep-04 Source Sep-04 S	55 05 05 05 05 05 05 05 05 05 05 05 05 0	4 Jan-05 Feb-0	and out that the the	Curr Yea Bud (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,000,000 1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Source States Sep-04 Source Sep-04 S	55 05 05 05 05 05 05 05 05 05 05 05 05 0	4 Jan-05 Feb-0	and out that the the	Curr Year Bud (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Project Li Sep-04 Project Li Sep-04	fe South of the south of the s	4 Jan-05 Feb-0	Por Source of the source of th	Curr Year Budg (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 Apr-04 May-04 Jun-04 Apr-	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Project Li Sep-04 Project Li Sep-04	fe South of the south of the s	4 Jan-05 Feb-0	Por Source of the source of th	Curry Year Budg (Curr
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Project Li Sep-04 Project Li Sep-04	fe South of the south of the s	4 Jan-05 Feb-0	Por Source of the source of th	Curry Year Budg (Curr
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 33,000,000 2,500,000 2,500,000 1,500,000 1,000,000 E0 E0 F0 F0 F0 F0 F0 F0 F0 F0 F0 F	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Project Li Sep-04 Project Li Sep-04	fe South of the south of the s	4 Jan-05 Feb-0	Por Source of the source of th	Curry Year Budg (Curr
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,000,000 2,000,000 1,500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Sep-04 Oct-04 Project Li Additional Sector	fe fe fe fe fe fe fe fe fe fe	4 Jan-05 Feb-0	A Start that the p	Droject has
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 1,500,000 E500,000 E500,000 E0 Base bar of key Points and suggestions a work has now ceased in light of the reference w been aborted. In the sevel of the means of extracting values we proceeding with urgent priority.	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Sep-04 Oct-04 Project Li Additional Sector	fe fe fe fe fe fe fe fe fe fe	4 Jan-05 Feb-0	A Start that the p	Curr Year Budg (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 Apr-04 May-04 Jun-04 Apr-	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Sep-04 Oct-04 Project Li Sep-04 Oct	Nov-04 Dec-04  fe  fe  shave been closed o investment made in  All external advisor c	Jan-05 Feb-0	A Start that the p	Curr Yea Bud (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 Apr-04 Apr-04 May-04 Jun-04 Apr-	Jul-04 Aug-04	Sep-04 Oct-04 Project Li Project Li Sep-04 Oct-04 Project Li Sep-04 Oct	Nov-04 Dec-04  fe  fe  shave been closed o investment made in  All external advisor c	Jan-05 Feb-0	A Start that the p	Curr Year Budg (Cur
E1,000,000 E500,000 E0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,000,000 1,500,000 E500,000 E500,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04  Project Li  Project Li  Source State St	Nov-04 Dec-04  fe  fe  shave been closed o investment made in  All external advisor c	4 Jan-05 Feb-0	A Start that the p	Curr Year Budg (Cur

## TRS00018615\_0121



Report for Month Ending:	28-Fe	eb-05	Pro	ject Manager:	Ken	McLeod
Start Date:	17-De			End Date:		
Overall Progress Status	17-D		ure 2004/5	Enu Date.	Project	Life Funding
overan i rogress Status		Expendit	ure 2004/5		Project	Life Funding
Progress Key:	1. 30000				Finance Key	:
On track for successful completion as programmed.					Within 10% of	
Issues have arisen which may delay completion or require		e discussion/direction.			10-20% outside estimate	
ssues have arisen which will delay com					>20% outside	eestimate
		Original Start	Original	Revised	Progress	Progress Sta
Critical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
Approval of appointment by FETA Boa	rd	17-Dec-04		Tana and States	С	
nception Report to FETA		17-Dec-04	12-Jan-05		С	
assist FETA with appointment of Tech		17-Dec-04	14-Jan-05		С	
ssist FETA with appointment of Lega		17-Dec-04	28-Jan-05		С	
repare LTS, AiP and draft Order for FI	ETA Board	17-Jan-05	3-Mar-05		IP	
ign Project Agreement with FETA		12-Jan-05	4-Mar-05		IP	
				11		State Provide H
				and a second second second		
			0.1.1.10	64. · · · · ·	C	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
	<b>F</b>		Original Cost	Start of Year	Current	N
the second s	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
revious Years	£0		£0	£0		£0
004/5	£26,000		£26,000	£26,000	£26,000	
005/6	£0		£0	£0		-£114,462
006/7	£0		£0	£0		-£9,538
uture Years	£0	£0	£0	£0		£0
otal for Project Life Cycle	£26,000	£26,000	£26,000	£26,000	£150,000	-£124,000
£30,000 T		2004/5				Actua
£25,000						oreca
						Oret
						Cost
£20,000						Cost (Cum
£20,000 £15,000				/		Contraction of the second s
				/		(Cum
£15,000				/		(Cum 
£15,000 £10,000						(Cum 
£15,000 £10,000 £5,000	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	15 Mar-05	(Cum 
£15,000 £10,000 £5,000 £0	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif		Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £0 £0 Apr-04 May-04 Jun-04	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £40,000 £40,000 £20,000	Jul-04 Aug-04			Jan-05 Feb-0	5 Mar-05	(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £60,000 £0 £0 £0 50 50 50 50 50 50 50 50 50 5		Project Lif	ie			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £60,000 £0 £0 £0 50 50 50 50 50 50 50 50 50 5		Project Lif	ie			(Cum 
£15,000 £10,000 £0 £0 Apr-04 May-04 Jun-04		Project Lif	e souther souther the souther the			(Cum 
£15,000 £10,000 £0 £0 Apr-04 May-04 Jun-04	A Jon Co Do	Project Lif	ie South of the south of the s			(Cum Year Budg (Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	A Jon Co Jon Co And A Harr	Project Lif	ie South of the south of the s			(Cum Year Budge (Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £60,000 £60,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	A Jon Co Jon Co And A Harr	Project Lif	ie South of the south of the s			(Cum Year Budge (Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	A son to	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	A port of the state of the stat	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	A port of the state of the stat	Project Lif	ie South of the south of the s			(Cum Year Budge (Cum
£15,000 £10,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 100,000 £80,000 £60,000 £40,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum)
£15,000 £10,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £20,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum)
£15,000 £10,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £20,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 120,000 £80,000 £80,000 £20,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum Year Budge (Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £00,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum Year Budge (Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £00,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,000 £80,000 £80,000 £20,000 £0 £0 £0 £0 £0 £0 £0 £0 £0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 
£15,000 £10,000 £5,000 £0 Apr-04 May-04 Jun-04 140,000 120,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum
£15,000 £10,000 £0,000 £0,000 20,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum
E15,000 E10,000 E10,000 E0 E0 Apr-04 May-04 Jun-04 40,000 20,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	ested course o d with FETA.	Project Lif	ie			(Cum
£15,000 £10,000 £0,000 £0,000 20,0	ested course o d with FETA.	Project Lif	ie South of the south of the s			(Cum 

۶

<b>Report for Month Ending:</b>	28-Feb-05	a distante internet	Pro	ject Manager:	Lindsay M	urphy
Start Date:				End Date:		
Overall Progress Status		Expendit	ure 2004/5		Project	Life Funding
			102%			
Progress Key: On track for successful completion as	n no group ad				Finance Key Within 10% c	
ssues have arisen which may delay con		discussion/direct	tion		10 - 20% out	
ssues have arisen which may delay con ssues have arisen which will delay cor		uiscussion/uneco			>20% outside	
		Original Start	Original	Revised	Progress	Progress Sta
Critical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R)
. Guideway Design		20-Jan-03	27-Jun-03	15-Sep-03	С	
. Acceptance of target cost		27-Jun-03	11-Jul-03	3-Nov-03	С	
. Guideway Construction		11-Jul-03	24-Mar-05	22-Nov-04	С	
. On Street Preliminary Design		5-Aug-02	7-Feb-03	25.0.101	С	
TROs		7-Feb-03	6-Feb-04	25-Oct-04	C C	
. On Street Detailed Design . Appoint On Street Contractor		7-Feb-03 10-Mar-03	1-Oct-03 1-Oct-03	22-Apr-04	C	
. On Street Construction		13-Oct-03	24-Mar-05	16-Jan-05	IP	
0. Driver Training		11-Nov-04	24-Mar-05	8-Dec-04	C	
1.Buses Operating for Public		24-Mar-05	24-Mar-05	9-Dec-04	C	
and the second second second second			011116	St	6	
	Funding	Budget	Original Cost Estimate	Start of Year	Current	Variance
revious Years	Funding £2,273,022	Budget £2,273,022	£2,273,022	Cost Estimate	£2,273,022	
nevious Years	£2,273,022 £7,959,694	£2,273,022 £7,959,694	£7,959,694	£2,273,022 £7,959,694		
005/6	£299,931	£299,931	£299,931	£299,931		£123,026
006/7	£0	£0	£0	£0		£0
uture Years	£0	£0	£0	£0	£0	£0
otal for Project Life Cycle	£10,532,647	£10,532,647	£10,532,647	£10,532,647	£10,532,647	£0
£7,000,000           £6,000,000           £5,000,000           £4,000,000           £3,000,000           £2,000,000           £1,000,000						(Curr 
E6,000,000 E5,000,000 E4,000,000 E2,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 12,000,000 26,000,000 E4,000,000 E4,000,000 E4,000,000 E5,000,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	Sep-04 Oct-04 Project Lif	Nov-04 Dec-04	4 Jan-05 Feb-0	15 Mar-05	(Cun —≜— Curre Year Budg
E6,000,000 E4,000,000 E4,000,000 E3,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 12,000,000 E8,000,000 E8,000,000 E0 E4,000,000 E0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	దీం రా సారా రా క్రిహ్ స్ట్రా రా రా క్రిహ్ స్ట్రా స్ట్రాల్ రా Lifetime Budget (C Actual/Forecast Co	mu)	4 Jan-05 Feb-0		(Curr Year Budg (Curr
E6,000,000 E4,000,000 E3,000,000 E1,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 12,000,000 E3,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E4,000,000 E0 Apr-04 May-04 Jun-04 May-04	Jul-04 Aug-04	ດ້ <sup>ຍ</sup> ູ້ເ <sup>ເ</sup> ັ້ນ ເວັ້ນ ເຈົ້າ ເວັ້ນ ເວັ້ນ ເມີດຍາຍ Lifetime Budget (C Actual/Forecast Co f action:	sé dé sé sé sé sé ter çé sé se ce ter um) ost (Cum)		or to be to	(Curr Year Budg (Curr Year (Curr Year Budg (Curr
£6,000,000 £3,000,000 £4,000,000 £1,000,000 £1,000,000 £0 Apr-04 May-04 Jun-04 12,000,000 £3,000,000 £3,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £5,000,000 £4,000,000 £5,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £4,000,000 £0,000,000,000 £0,000,000	Jul-04 Aug-04	مجمع محمد من معلم من معلم من محمد من محمد من محمد من محمد من معلم من محمد من محم محمد محمد محمد من محمد من محمد من محمد من محمد محمد	ی می	ext few weeks. Disc ks and are continui pensation events f as been reallocated ameras and site sup	and the source of the source o	(Curr Year Budg (Curr Year Budg (Curr Solo Solo Solo Solo Solo Solo Solo Solo
E6,000,000 E5,000,000 E4,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 Ju	Jul-04 Aug-04	م م م م م م م م م م م م م م	so the set of the set	ext few weeks. Disc ks and are continui opensation events f as been reallocated ameras and site sup int has been made to the existing network rks are invoiced as	ussions are uning to snag the for this contract into 2005/6 to opervision of our to account add k these were oper the agreed	(Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Budg (Cun Year Feo Station (Cun Year Feo Station (Cun Year Feo Station (Cun Year Feo Station (Cun Year) Feo St
E6,000,000 E5,000,000 E4,000,000 E1,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 12,000,000 E6,000,000 E6,000,000 E4,000,000 E6,000,000 E6,000,000 E0 E0 E0 E0 E0 E0 E0 E0 E0	Jul-04 Aug-04	by b	as as a start of the second area £120,000 has been second at area £120,000 has been second at area £120,000 has been forward to con a start area £120,000 has been for the CCTV con the area £120,000 has been second at area \$120,000	ext few weeks. Disc its and are continuin opensation events f as been reallocated ameras and site sup int has been made the existing network rks are invoiced as deration. Work is u	ussions are uning to snag the for this contract into 2005/6 to opervision of our to account add k these were oper the agreed	(Curr Year Budg (Curr Year Budg (Curr Solo Charl Composition Solo Charl Composition Compos
26,000,000 25,000,000 25,000,000 25,000,000 20,000	Jul-04 Aug-04	st signals and sig	As as as a set of the	ext few weeks. Disc ks and are continuinpensation events f as been reallocated ameras and site sup int has been made the existing network rks are invoiced as deration. Work is u	ussions are uning to snag the for this contract into 2005/6 to opervision of our to account add k these were oper the agreed	(Curr Year Budg (Curr Year Budg (Curr Solo Charl Composition Solo Charl Composition Compos
E6,000,000 E4,000,000 E4,000,000 E3,000,000 E1,000,000 E0 Apr-04 May-04 Jun-04 12,000,000 E8,000,000 E8,000,000 E0 E4,000,000 E0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	st signals and sig	as as a start of the second area £120,000 has been second at area £120,000 has been second at area £120,000 has been forward to con a start area £120,000 has been for the CCTV con the area £120,000 has been second at area \$120,000	ext few weeks. Disc ks and are continuinpensation events f as been reallocated ameras and site sup int has been made the existing network rks are invoiced as deration. Work is u	ussions are uning to snag the for this contract into 2005/6 to opervision of our to account add k these were oper the agreed	Budg (Cum

	t: Ingliston Park	anu Mue	D	ingt Man	I in dage M	manhar
Report for Month Ending			Pro	ject Manager:	Lindsay M	urphy
Start Date			the second second	End Date:		
<b>Overall Progress Status</b>		Expendit	ure 2004/5		Project	Life Fundin
			59%	1.4		
rogress Key:					Finance Key	
n track for successful completion a	s programmed.				Within 10%	
sues have arisen which may delay c		discussion/direct	tion.			side estimate
sues have arisen which will delay co		· · · · · · · · · · · · · · · · · · ·		in the property of	>20% outside	
		Original Start	Original	Revised	Progress	Progress St
ritical Path / Milestone Items		Date	Completion	Completion	(NS,IP,C)	(G,Y,R
ppoint Consultant		15-Aug-03	22-Aug-03		C	
ception Report to CEC		18-Sep-03	18-Sep-03		С	
etailed Design and Study Work		18-Sep-03	2-Jan-04		C	
etailed Planning Consideration (12	weeks)	2-Jan-04	26-Mar-04	30-Apr-04	С	
repare Tender Documentation		1-Dec-03	5-Mar-04	12-Mar-04	С	
ender Period		10-Mar-04	20-May-04	12-Jul-04	C	
onstruction		21-May-04	3-Jan-05	10-Jun-05	IP	1
and the second						
	Sec. 2					
			Original Cost	Start of Year	Current	
	Funding	Budget	Estimate	Cost Estimate	Forecast	Variance
revious Years	£106,417		£106,417	£106,417	£106,417	£0
004/5	£2,469,465	£2,469,465	£2,469,465	£2,469,465		£1,010,354
005/6	£0	£0	£0	£0		-£1,506,998
006/7	£0	£0	£0	£0	£0	£0
uture Years	£0	£0	£0	£0		£0
otal for Project Life Cycle	£2,575,882	£2,575,882	£2,575,882	£2,575,882	£3,072,526	-£496,644
£2,000,000 £1,500,000 £1,000,000 £500,000		***			-	Cos (Cur ————————————————————————————————————
£2,000,000	Jul-04 Aug-04	Sep-04 Oct-04	Nov-04 Dec-04	4 Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
E2,000,000 E1,500,000 E1,000,000 E0 E0 Apr-04 May-04 Jun-04	Jul-04 Aug-04	and a model		Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
£0 Apr-04 May-04 Jun-04	Jui-04 Aug-04	Sep-04 Oct-04 Project Li		Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
£2,000,000 £1,500,000 £1,000,000 £500,000 £0	Jul-04 Aug-04	and a model		Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
£2,000,000 £1,500,000 £1,000,000 £0 £0 Apr-04 May-04 Jun-04 3,000,000	Jul-04 Aug-04	and a model		Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
£2,000,000 £1,500,000 £1,000,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000	Jul-04 Aug-04	and a model		4 Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
E2,000,000 £1,500,000 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000	Jul-04 Aug-04	and a model		4 Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
£2,000,000 £1,500,000 £1,000,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,000,000 1,500,000	Jul-04 Aug-04	and a model		4 Jan-05 Feb-0	5 Mar-05	Cos (Cur ————————————————————————————————————
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 4,500,000 1,000,000 £0 4,500,000 1,000,000,000 1,000,000,000 1,000,000 1,000,000 1,000,000		Project Li				Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 4,500,000 1,000,000 £0 4,500,000 1,000,000,000 1,000,000,000 1,000,000 1,000,000 1,000,000		Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 4,500,000 1,000,000 £0 4,500,000 1,000,000,000 1,000,000,000 1,000,000 1,000,000 1,000,000	A cold saturday and the second	Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0 £500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 500,000 £0 500,000 £0 500,000 £0 500,000 500,000 £0 500,0000 500,000 500,000 500,000 500,0000 500,000 500,000	A por the training of the trai	Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 1,500,000 5500,000 1,000,000 2,500,000 1,000,000 2,500,000 1,000,000 2,500,	Acch in the participation	Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 2,500,000 1,500,000 5500,000 1,000,000 2,500,000 1,000,000 2,500,000 1,000,000 2,500,	Acch in the participation	Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 £1,500,000 £0,000,000,000 £0,000,000,000 £0,000,000,000,000,000,000,000,000,000,	Acch in the participation	Project Li	fe			Cos (Cur Yea Bud (Cur
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	A CON INT COST AND	Project Li	fe	and that existing sign	A OC A C OC Y	Cos (Cur Yea Bud (Cur (Cur Yea Bud (Cur (Cur (Cur)) (Cur (Cur)) (Cur (Cur)) (Cur) (Cur)) (Cur)
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	A ch ch course of ggested course of with City of Edinburg bhling designs have be	Project Li	fe	And wat existing signation and Border Co	s be rationalise nstruction now	Cos (Curr Yea Bud (Curr Yea (Curr Y (Curr Y (Curr Y (Curr Y (Curr Y (Curr (Curr Y
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	A ch ch course of ggested course of with City of Edinburg bhling designs have be	Project Li	fe	And wat existing signation and Border Co	s be rationalise nstruction now	Cos (Cur Yea Bud (Cur (Cur Yea Bud (Cur (Cur (Cur)) (Cur (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur)) (Cur) (Cur)) (Cu
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	A ch ch ch ch ch ch ch bod yet ce ch yet bot ch ggested course of with City of Edinburg phting designs have be stifield Road and a part is the 7 <sup>th</sup> June.	Project Li	fe	ng that existing signation and Border Co	A OC A A OF Y	Cos (Curr Yea Bud (Curr Yea (Curr Y (Curr Yea (Curr Y (Curr Y (Curr Y (Curr Y (Curr Y
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	A Ch City of Edinburg ggested course of with City of Edinburg shing designs have be stifield Road and a par is the 7 <sup>th</sup> June. ng and CCTV cable d 80% complete. Kerb	Project Li	fe fe fe fe fe fe fe fe fe fe	ng that existing sign ction and Border Co Consultation will co	s be rationalise nstruction now port source on 1.	Cos (Cur Yea Bud (Cur (Cur Yea (Cur Yea (Cur Yea (Cur Yea (Cur Yea (Cur (Cur Yea (Cur (Cur (Cur)) (Cur (Cur)) (Cur (Cur)) (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur)) (Cur) (Cur)
E2,000,000 £1,500,000 £0,000,000,000 £0,000,000,000,000,000,000 £0,000,000,000,000,000,000,000,000,000,	A Ch City of Edinburg ggested course of with City of Edinburg shing designs have be stifield Road and a par is the 7 <sup>th</sup> June. ng and CCTV cable d 80% complete. Kerb	Project Li	fe fe fe fe fe fe fe fe fe fe	ng that existing sign ction and Border Co Consultation will co	s be rationalise nstruction now port source on 1.	Cost (Cur Year Budy (Cur Year Budy (Cur Year Budy (Cur Year Budy (Cur Year Budy (Cur
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 May-04	Bested course of with City of Edinburg phing designs have be stifield Road and a part is the 7 <sup>th</sup> June. Ing and CCTV cable d 80% complete. Kerb mpleted. Scottish Wat	Project Li	fe b b b c c c c c c c c c c c c c	ang that existing sign ction and Border Co Consultation will co further progress to s area. Roundabout co external blockwork h	s be rationalise nstruction now commence on 1 sub base to acc completed. Uti as been compl	Cos (Cur Yea Bud (Cur (Cur Yea Bud (Cur (Cur Yea Bud (Cur (Cur)) (Cur) (Cur)) (Cur) (Cur)) (Cur)) (Cur) (Cur))
22,000,000 21,500,000 2500,000 20 20 20 20 20 20 20 20 20	a construction of the second s	Project Li	fe	ng that existing sign ction and Border Co Consultation will co further progress to s area. Roundabout of external blockwork h ays incurred to the ng predictions Risk if	s be rationalise nstruction now completed. Uti as been completed. Uti as been completed. Uti as been completed. Uti	Cos (Cur Yea Bud (Cur (Cur Yea (Cur Yea (Cur Yea Bud (Cur Yea (Cur (Cur (Cur (Cur (Cur (Cur (Cur (Cur
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04	Be charter of the second secon	Project Li	fe	ng that existing sign ction and Border Co Consultation will co further progress to s area. Roundabout of external blockwork h ays incurred to the ng predictions Risk if has been arranged	s be rationalise nstruction now completed. Uti as been completed. Uti as been completed. Uti as been completed. Uti	Cos (Cur Yea Bud (Cur (Cur Yea (Cur Yea (Cur Yea Bud (Cur Yea (Cur (Cur (Cur (Cur (Cur (Cur (Cur (Cur
E2,000,000 E1,500,000 E1,500,000 E0 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Apr-04 May-04 Jun-04 Jun-04 May-04 Jun-04 Jun-04 Jun-04 May-04 May-04 Jun-04 May-04 May-04 May-04 May-04 Jun-04 May-04	Be charter of the second secon	Project Li	fe s sisues. Due to del nec covers the existing t progress and fina	ng that existing sign ction and Border Co Consultation will co further progress to s area. Roundabout of external blockwork h ays incurred to the ng predictions Risk if has been arranged	s be rationalise nstruction now completed. Uti as been completed. Uti as been completed. Uti as been completed. Uti	Cos (Curr Yea Bud (Curr Yea (Curr Y (Curr Yea (Curr Y (Curr (Curr Y (Cur
E2,000,000 £1,500,000 £0 Apr-04 May-04 Jun-04 3,000,000 2,500,000 1,500,000 £0 4,500,000 1,000,000 £0 4,500,000 1,000,000,000 1,000,000,000 1,000,000 1,000,000 1,000,000	Be charter of the second secon	Project Li	fe	ng that existing sign ction and Border Co Consultation will co further progress to s area. Roundabout of external blockwork h ays incurred to the ng predictions Risk if has been arranged	s be rationalise nstruction now completed. Uti as been completed. Uti as been completed. Uti as been completed. Uti	ed and the new have approval. 4 <sup>th</sup> March 2005 cess roads and lity diversion in eted. Terminal e the predicted ngency for final





Agenda Item 8b

# **Other Projects**

b) Ingliston Park & Ride – Lessons Learnt \*

tie Limited



tie board meeting 21<sup>st</sup> March 2005 Ingliston Park and Ride Lessons Learnt

### Steering Group

A steering group meeting has been arranged for the 17th of March to discuss and progress all aspects of the delivery of the project. This will be attended by representatives from Transport, **tie**, Halcrow, and bus operators.

### Construction

Construction work is generally progressing well and to a good standard Utility Diversions have now been arranged and in the mean time work has been rearranged around the area to minimise delay. The building is taking shape with the roof at an advanced stage and the blockwork complete. The recent inclement weather has caused some disruption however this is being managed. Surfacing of the car park commences this week.

Scottish Power have caused some further problems as they had initially intimated that they would supply the site from cables in Eastfield Road. They have now intimated that this will not be the case and the contractor is awaiting their proposal for an alternative source. The contractor is preparing a plan to mitigate any further delay that this would cause to the delivery of the site.

In addition Street Lighting have caused some delay in responding with requests for further information.

### **Fitting Out**

A schedule of colour and fitting choices has been received which will require instruction. In addition provision of equipment not supplied by the contractor e.g. Vending Machines, computer terminals and telephones requires consideration.

#### Traffic Regulation Orders and Parking Places Order

Traffic Regulation Orders and a Parking Places Order are required to enforce bus lanes, waiting restrictions, turning restrictions and the regulation of the car park. The formal consultation period for these will commence this week following which the orders will be advertised. If objections are received then a report will be required to go to the 7th June Council Executive Committee.

### Operation

The City's officers are at an advanced stage in the preparation of operating agreements with bus operators who have stated an interest in serving the site this includes staffing the building and light maintenance. **tie** will continue to provide support.





### Training

Near the end of the construction period staff will require to be trained in the use of any equipment provided for the operation of the site and building. This has been added to the overall programme.

### Launch

The date for completion of the Construction is presently the 10th of June. It has been suggested that the P.R. Launch be aligned to the completion of Hermiston Park and Ride and the Royal Bank of Scotland opening in July However cognisance needs to be taken of the G8 conference which is taking place in the first two weeks in July.

This will be discussed at the Steering Group. Some thought has been given to two options for the opening of the Ingliston site. One invitation only and one for members of the public (the Bouncy castle approach) these have been costed and a draft programme for delivery produced.

### Project Management and Control

The form of contract with the contractor for Ingliston is the Engineering and Construction contract option A which is design and build with a priced activity schedule. The reference design was prepared by our client's representative pre tender and the client's representative is responsible for all quality assurance issues and cost certification. The contractor is responsible for the delivery of a robust product with the design being certified by the contractor's designer. In the case of Ingliston, the client's representative is Halcrow, the contractor is Borders Construction and the contractor's designer is Goodson Associates.

The role of the **tie** project manager in this context is very much one of monitoring the client's representative and providing the necessary interface with the city council and other third parties to provide a total project.

#### Lessons from WEBS

The board will recollect a paper presented to the meeting on 20th December that addressed lessons to be learned from the procurement of WEBS with a view to ensuring that those lessons are imported into other **tie** projects. The process for capturing the lessons that was outlined to the Board in December is as follows.

The process of capturing the lessons to be learned involves four stages. Stage 1 was a first stage issues review involving the projects director and the two project managers that were involved in the scheme. Some headline results of that exercise are included later in this paper.



The second stage will involve sessions with the two main construction contractors Balfour Beatty and ERDC. There are a number of significant contractual issues to be resolved with the contractors regarding quality of product, outstanding work to be completed, programme to final completion and financial issues.

The third stage will involve a similar review with our client representatives, Halcrow. Again there are a number of issues to be resolved with them including quality of service, financial issues and programme to completion.

The fourth stage will involve a review with Lothian Buses and CEC regarding the interface between these parties and **tie** and how, with the benefit of hindsight, we could have improved.

Finally the results of this will be fed into the business improvement work that is ongoing to ensure that our project management procedures benefit from the experience.

At present stages 1 and 4 are complete and stages 2 and 3 are ongoing with continued work to resolve outstanding contractual issues. However there are some conclusions of the review that are directly applicable to the Ingliston context and that have been implemented. The relevant conclusions are as follows.

• The level of interface with CEC and other bodies such as Lothian Buses involves a greater degree of commitment than originally envisaged

• The degree of supervision of the client's representative in technical areas like quality, programme and price has proven to be greater than expected. The incentive for **tie** staff to look after the interests of **tie** is greater than that of the client's representative.

• The degree of site supervision required directly by **tie** to ensure a quality product has proven to be greater than expected.

• It is essential to have a tight control over the contractual issues related to the client's representative as well as the contractor. There is always potential for disputes between the two parties as to liability when defects are discovered.

• As a result of the above the level of project management resources we need on projects at the construction stage needs to be increased.

#### Actions taken on Ingliston

As a result of the review of WEBS a number of actions have been taken to ensure that Ingliston goes as smoothly as possible.

 An additional project manager has been redeployed to Ingliston on a part time basis to assist

#### tie Limited



• Input has been provided from our QS and tram programme manager to assist with valuations, contract negotiations and programme

• Input has been provided from the tram PU manager to assist the contractor in negotiations with PU undertakers.

 An experienced clerk of works has been deployed to ensure that the client's representative site supervision is adequate.

As a result of these actions considerable progress has been made in clarity of programme, valuation of early warning notices for compensation events, PU diversions and traffic regulation orders, all of which are potentially significant issues for this contract. It is important to recognise that these additional resources are deployed to safeguard tie's interests and provide the necessary clarity in the event of a dispute arising with either the contractor or the client's representative. In the context of potential settlement levels for compensation events, the contingency allowance for which currently stands in the business plan at £500,000, this level of input is expected to deliver value for money.

Report by Lindsay Murphy/Alex Macaulay March 2005



2

Agenda Item 8c

## **Other Projects**

c) ITIBS \*



## Following the NO vote: Realising value from congestion charging development

1: IT and business systems

CONFIDENTIAL

#### Introduction

**tie** has undertaken innovative work in developing a congestion charging scheme for Edinburgh, including procuring and partnering the development of IT systems required to implement such a scheme.

This paper examines areas where the experience and expertise gained by **tie** in the IT aspects of this project could potentially be exploited following the 'no' referendum vote. It examines how the ITI Business System ('ITIBS') developed to run the charging scheme could be used for alternative appropriate applications in Edinburgh or elsewhere in the UK, ensuring that the inherent value built up for the charging scheme is not lost.

A separate note deals with potential opportunities in supporting other aspects of congestion charging scheme design and statutory processes at other locations.

### The ITI Business system - ITIBS

As part of the development stage for the congestion charging scheme, detailed design work was undertaken on systems to operate the scheme, following an innovative procurement strategy. The approach used for this work, the systems developed and the experience gained are relevant to the wider use of IT systems for transport applications, and potentially in other public sector areas.

They have the capability to underpin the delivery of major advances in integration, efficient traffic management, user-friendliness and quality for transport users of all kinds – objectives set out in the Scottish Executive's 2004 White Paper and reinforced by Alastair Darling in his Scotsman article of 23 February 2005. In England, a substantial 'Transport Innovation Fund' has been established to support new approaches to tackling these objectives. These issues will be critical in reporting on alternative approaches to controlling increasing congestion as required by the City of Edinburgh Council in their decision of 24 February.

#### The evidence

The system procurement for ITIBS was driven by a number of objectives:

- Minimising 'total cost of ownership' total capital and operating costs
- Minimising delivery programme risks
- · Avoiding long-term reliance on a single supplier
- Maximising opportunities to add value.

The key to achieving these was a focus on business processes.

Contracts were let to two suppliers, stage 1 of which was to undertake detailed design work for the charging system based on the objectives above.

This work was successfully completed and prototypes demonstrated by the contractors on time and to budget. Proven features are:

The system developed for **tie** is very substantially cheaper than that in place for London's charging scheme and is within the constrained budget allowed for in the charging scheme business case.

The system relies on standard components, minimising interface issues and avoiding the use of bespoke software that only a single supplier would be able to support.

The system is fully flexible and scalable and can easily allow for additional modules to provide additional functionality.

Business processes and technology principles reflecting best practice in the private sector have been designed into all areas of transaction, analysis, and reporting covering the entire operational spectrum of a business operation.

The core of the system is a 'back office' using industry standard software (Oracle, SAP) that can handle any financial and/or data management requirement and can be readily updated and adapted.

#### The business system

The ITIBS concept is based on the design of business processes that can then be used for detailed technical specification – particularly for the configuration of standard software such as Oracle, and the incorporation of any interfaces with external applications. The framework is set out in the diagram below:



The business processes, while required for congestion charging, are generic and can be applied to other uses. Looking at each of these in turn:

- Registration: account information, eg blue badge issue, residential parking permits.
- Work flow: management, programming and allocation of incoming workstreams, eg customer care, mail management, telephone calls.
- Escalation: conditional follow-up action based for example on performance standards or service level agreements, eg debt management, penalty notices, customer care.
- Matching processes: for reconciliations /comparisons, eg parking offenders
- Channel management: sales processes including web, sms and interfaces with retail outlets.
- Financial management: self explanatory
- Public and Management information: based on all the information held by the system, with significant ability to cross reference, eg real time information, parking guidance.

The features of the system developed would be best exploited in promoting integrated systems where data exchange between differing applications can add significant value beyond that provided by applications in isolation (in other words situations where the whole is more than the sum of the parts).

In the transport field, UTMC aims to facilitate such integration for traffic control, information and management application. It specifies standards for applications in this field to allow for the possibility of such integration to take place.

The **tie** approach is not inconsistent with this, but starts with a central processing platform that will give powerful analysis and management of information, including the ability to share it in various forms (eg web, SMS) and the possibility to run entire operations (including channel offerings and full financial management). A whole range of applications can be supported by using such core business engine, not simply transport-related ones.

The difference is illustrated in the diagram below:



### Potential tie role

The experience gained could be applied to IT procurement for other transport or related public sector projects. **tie** could act as either:

- 'Intelligent client' acting in partnership with a public sector project promoter (similar to tie's role with FETA on the statutory processes); or
- Consultant advising a public sector project promoter on a commercial basis.

As 'intelligent client', **tie** would provide unique experience of best practice from both public and private sectors with a small team able to engage directly with IT delivery organisations. The NAO report on "Improving IT procurement", 5 November 2004, specifically recommends such an "intelligent customer" approach in realising success and value from IT enabled projects

As adviser, **tie** would be able to offer specialist advice on all aspects of procurement ranging from selection of tenderers to end user acceptance programmes.

Of these, the role of 'intelligent client' is preferred, on the grounds that the experience gained and the skills available within **tie** are most suited to providing the bridge between supplier and promoter and ensuring project delivery is focused on promoter needs.

There is a further possibility, which is that **tie** could provide an IT service to Councils in relation to some of the areas described below. This would be a modest parallel to a Danish organisation that provides IT services to local authorities on a very large scale: KMD plc. This company is owned by a consortium of Danish public authorities.

### Potential applications

The ITIBS business system lends itself to a wide range of applications in both the transport field and the wider public sector. Indeed the more applications are linked to the system, the greater are the opportunities to benefit from synergies in information management, customer payment channels, and customer relationship management tools that they may be unlikely to be able to support on an individual basis. Most importantly, the consolidation of information into one central engine allows wide scope to achieve integration and other key service objectives that would otherwise be very difficult or impossible to realise.

The key to delivering the benefits is to realise the synergies from this 'shared service' supporting a wide range of projects. The costs would be similarly shared over a range of projects. However, it is likely that a project of a significant size will be needed to support initial implementation. This could then be extended into other, possibly smaller, applications at relatively low marginal cost.

In the Edinburgh context, it is anticipated that the most likely core applications would be focused around **traffic management and control** systems as a tool for congestion management. These systems will also be required for effective management of the tram system, and even more importantly for the tram construction period. Initial work on developing an integrated urban traffic management and control (UTMC) strategy has been carried out for CEC by Halcrow, and the Council is seeking funding to develop this further, part of which may come from the tram implementation funding package.

The ITI business system could potentially provide a significant contribution to such a strategy. It could link traffic signal control over a wide area with public transport real time information, the provision of timetable information and personalised journey planning, parking information systems and real time roadside information about travel delays etc. This is potentially a very large area with substantial opportunities for innovation and demonstration of best practice.

To a core application such as this could then be added a whole range of additional uses supporting Council activities. These might include:

Parking

The whole range of parking management activities including payment processes, enforcement activities, permit and exemption registration, contract management and user information systems could be covered by ITIBS. This operation is already a self-contained activity within CEC which is currently dependent on bespoke IT processes that are now outdated and not capable of direct upgrading. Incorporating parking management into ITIBS would allow parking information to be applied for wider traffic management and control purposes.

#### Tram Back Office

The tram system will need a back office providing financial control, management of payment channels and ticketing, etc. ITIBS has again got the potential to provide this facility. While initial discussions with Transdev suggest that the requirement in this respect will be quite limited, it might still make a viable marginal application for ITIBS, particularly if linked with integrated ticketing (see below).

#### Road network asset management

CEC aspires to a more strategic approach to managing the maintenance of the road network, while the Scottish Executive is including powers for the coordination and management of roadworks by utilities in the current Transport Bill. Systems allowing the prioritisation of maintenance works against predetermined criteria, information provision to road users and local authorities, and coordination issues could be very effectively handled by ITIBS. There may well be further areas under this heading that could also benefit.

#### Air quality

Without congestion charging, alternative approaches may be needed to meet the targets of the Council's statutory air quality action plan. Linking air quality monitoring into ITIBS could allow automatic initiation of mitigation measures during episodes of poor air quality. This might for example include alternative signal plans and variable message signs at appropriate locations – applications that would also be part of the overall ITIBS package.

#### Integrated ticketing

The system would provide the facilities required if the One-ticket arrangements were to be scaled up into a major operation (and including the tram ticketing arrangements). The business processes for this would be very similar to congestion charging in terms of purchasing of tickets/licences through a wide range of payment channels, payments to operators and other creditors, with secure accounting and management information processes.

There are also likely to be opportunities in some of the above applications with other Councils, or, in future, Regional Transport Partnerships. However, a strong core application is still likely to be needed to form the basis of such a shared service. There are further applications that could be relevant at a national level:

#### Smartcard and/or concessionary travel arrangements

The Scottish Executive are developing national concessionary travel arrangements, which will be based on smartcards held by eligible users. Such smartcard arrangements could be extended to other types of users. These could include for example other travel concessions, integrated ticketing for any traveller, or even payments for other goods as is now being considered for the London Oystercard system. ITIBS would be able to provide all the back office management systems for such an arrangement. Blue badge registration and management

Registration of blue badges is currently undertaken by individual local authorities with no compatibility or standards for databases, or even of criteria for issue. However, the ability to check the validity of badges is likely to be increasingly required on a national basis. ITIBS could provide a service to local authorities throughout Scotland for issuing badges and maintaining a comprehensive database.

Speed limit enforcement

Speed cameras based on measuring vehicle times over a significant length of road ('SPECS') are being proposed in Scotland (Scotsman 3 March), in addition to the 'single point' speed cameras already in place. The business processes required for these systems to process enforcement action closely match those required for congestion charging, and are therefore well within the scope of ITIBS.

Congestion charging application elsewhere in the UK

This would supplement the wider role that **tie** could provide to authorities considering charging schemes discussed in a separate paper. It is therefore proposed that this be packaged with the other aspects of charging scheme development as part of a 'tie offer'. Discussions should be held with those authorities seriously considering charging, and with London which is seeking to extend and develop its existing scheme, to identify potential areas for IT collaboration.

The 'shared service' approach would allow for a structure where local councils would only need to deploy on-street technology, and the resources required for front and back office processes could be shared. Such an approach would lower total costs through economies of scale.

The **tie** team believe there is potential for ITIBS to provide benefits in both financial and cost-benefit terms for Edinburgh - and potentially on an even larger scale if applied Scotland-wide. However, a quantitative evaluation of such benefits has not been carried out at this stage, nor have funding sources been examined. **tie** recognise that a satisfactory business case would have to be made in proposing to take any further proposals forward.

#### Next steps

tie propose:

 To supplement the major investment to date in development of the ITIBS Business System by preparing a preliminary business case for its further application. This would aim to identify where value could be realised from the investment made despite the rejection of Edinburgh's congestion charging proposals in the referendum. It would include the role of tie in this process and would cover:

- Scoping of potential uses with particular focus on Edinburgh, using traffic control and management as the starting point;
- Examining the potential for regional applications in SESTRAN context;
- Examining the potential for supporting a national strategy for transport IT development and application;
- Outline costing of applying the system to such uses;
- Examining opportunities for establishing a 'shared service provider', including the relevance of the Danish KMD model;
- Initial assessment of potential benefits in both financial and costbenefit terms;
- Identification of funding options for implementation;
- Risk assessment;
- Comparison with existing standards or 'conventional' methods (eg UTMC);
- Relationships with the existing Council IT contract.
- To seek opportunities to support this and other IT activity by acting as 'intelligent client' with public sector bodies.

This work would be carried out by existing **tie** staff, with minimal third party support. If the approach is agreed, an interim report would be produced for the Board and CEC in May, with this stage of work being completed by September. Close liaison with the Scottish Executive will also be necessary. It is proposed that this should be funded from the residual budget for congestion charging development unspent as a result of the referendum result. This amounted to £160,000.



## BOARD MEETING 21 MARCH 2005

Following the NO vote: Realising value from congestion charging development

2: Scheme design and statutory procedures

CONFIDENTIAL

### Introduction

**tie** has undertaken innovative work in designing a congestion charging scheme for Edinburgh and taking it through all the consultative and statutory procedures.

This paper examines where the experience and expertise gained by **tie** in the development of the scheme could potentially be exploited following the 'no' referendum vote. A separate note deals with the potential opportunities arising from IT and business systems design.

#### Scheme design and statutory processes

Edinburgh's congestion charging scheme was ready for implementation subject to final confirmation and approval by Scottish Ministers. **tie** has successfully steered this scheme through the preliminary and detailed design, extensive informal consultation, Council decisions and the formal statutory procedures including a public inquiry.

The public inquiry – with a successful outcome – is the only one to have been held on a congestion charging scheme in the UK.

The experience gained could be used in two ways. Firstly **tie** could provide advice to other traffic authorities considering a charging scheme. The company has already been appointed by FETA to assist the project management of the replacement of tolls on the Forth Road Bridge with a road user charge under the 2001 Transport Act. This activity is likely to continue into 2007. **tie** could provide assistance to government(s) in developing regulatory frameworks for charging schemes elsewhere. Secondly, **tie** can contribute towards the review of transport strategy agreed by the Council on 24 February, including whether any consensus exists for alternative approaches to congestion management in Edinburgh.

#### Supporting schemes elsewhere

Outside Scotland the development of congestion charging in the UK will depend on the uncertain political appetite for such schemes both by Councils and by national government. In addition, legislation and approval requirements are different from those in Scotland. There may however be opportunities to provide an advisory role in cities considering congestion charging. Manchester and Cardiff, for example, are known to be examining charging options. There may also be opportunities outside the UK, although the details of statutory processes etc will have some differences from those in Scotland.

**tie** therefore should identify in some detail the areas in which it could provide information, support or advice to local authorities and national government. At the very least, a comprehensive analysis of the lessons learnt from the

Edinburgh experience should be undertaken. In summary, areas that could be examined include:

- Strategic considerations in designing a charging scheme;
- Using consultation and market research to assist design;
- Promoting acceptance;
- Stakeholder relationships;
- Decision-making processes;
- Forecasting and appraisal of impacts at appropriate levels of detail;
- Managing the statutory procedures, including public inquiry;
- The role of government guidance;
- Technology issues including procurement (dealt with in a separate paper)
- In relation to all these potential risks/barriers to success

It is proposed that a short analysis of what **tie** could offer in each of these (and any other relevant) areas should be prepared, together with a marketing strategy for exploiting this experience. This could for example involve **tie** seeking to identify and approach key senior officials within major UK cities and the Department for Transport with a **tie** proposal.

#### Edinburgh

The second way of using the experience gained would be to support any proposals for an alternative future charging scheme, or other methods of tackling congestion, for Edinburgh. Recent events have shown that there is serious concern about congestion in the city; and that for certain groups, an alternative charging scheme might be acceptable. Retailers groups and the Liberal Democrats for example have both indicated they are not against charging in principle. Clearly there is no possibility that any scheme could be considered prior to the 2007 elections, but it may be that third parties could put forward proposals for the longer term. The Council in its decision of 24 February agreed to receive reports from community groups on proposals for tackling congestion.

**tie** would be available to support any informal discussions with retailers groups or any other interested bodies to explore options for the future, but would not initiate such discussion.

#### Next steps

- To undertake an analysis of experience gained and expertise that could be used to assist local and national government in the UK (and potentially abroad) in developing congestion charging schemes and processes. This would then be linked to a scoping of marketing opportunities and a strategy for exploiting these.
- To participate as required in informal discussions with Edinburgh retailers groups and any other interested bodies to explore options for the future in tackling congestion in the city.

J Saunders 16 March 2005



Agenda Item 9

# Communications

- a) Information Programme \*
- b) Stakeholder Report \*



Agenda Item 9a

# Communications

## a) Information Programme \*



Transport Edinburgh Communications Strategy in March 2005 This note seeks to update the tie board of CEC's recent progress

Voters in the Transport Edinburgh referendum voted against the council's preferred strategy, which included the introduction of a congestion charging scheme. This means that the council will now follow its base transport strategy. This strategy is an ambitious programme that includes, two tram lines, new park & ride facilities at Ingliston, Hermiston, Straiton and Todhills, a record £28.4m in 2005 for road maintenance, the Central Edinburgh Traffic Management scheme (CETM) and the improvements for pedestrians in the city.

With this programme in mind the council is seeking approval for a revised transport communications strategy and is currently preparing a communication plan to support these activities.

The recent experience of the transport referendum has shown how crucial to the success of transport policy is a supportive body of public opinion. Interest in the council's transport policy will now focus on trams and the parking review. It is essential to the success of these projects that they enjoy a wide degree of public support. To this end the council, tie, and corporate communications is undertaking a planning exercise to formulate an integrated and coordinated strategy for transport.

The next edition of Outlook with a four-page transport supplement on the outcome of the referendum is now completed. This issue will be distributed on 25 March.

Sue Campbell

Mar 2005

CEC, Dec 04



Agenda Item 9b

## Communications

b) Stakeholder Report\*



Board Update Stakeholder & Communication Management 21 March 2005

Stakeholder management:

- Stakeholders
- Newsletter

Communication management:

- Media enquiries
- Website
- Trams Communication Planning
- One Ticket Marketing Plans
- Communication strategy and partners
- Events

#### Stakeholders

The following presentations were made in the last month:

Strategic Programme Partnership	9 March 2005
Railway Industry Association	10 March 2005

Following the result of the Congestion Charging referendum on 22 February communications are at a planning stage. Plans, including specifics on stakeholders, will be presented to Michael Howell and Alex Macaulay on 22 March for approval.

#### Newsletter

The Stakeholder Newsletter was issued on 28 February with updated information relating to the Congestion Charging result and project updates. An additional seven requests have been received to be added to distribution of the newsletter.

Proposals to pull the newsletter into an on line e-news format sourced from the **tie** website, with a page for each **tie** project, have been documented showing costs and benefits. Proposals will be sent to Michael Howell and Alex Macaulay for approval.

#### **Media enquiries**

A proactive press release from tie was issued immediately following the Congestion Charging referendum result announcement. The press release offered both tie comment and a question and answer segment to help the media and to reduce the amount of individual requests for quote and comment.

Interviews are planned with the following media:

#### 22 March 2005

Michael Howell meets with Alistair Dalton of the Scotsman.

#### 24 March 2005

Initial meeting to be held with the ESPC to look at the possibilities for a positive story relating to property prices around tram lines.

#### Website

The **tie** website was updated immediately to reflect the result of the Congestion Charging referendum result on 22 February.

#### **Trams Communication Planning**

A meeting on 9 March started the communication planning for tram lines 1 and 2. Attended by Barry Cross, Trams Project Managers, Stakeholder Management, Weber Shandwick and Communications the following has been agreed for lines 1 and 2:

#### By 15 March 2005

1. A positive story based on the work being progressed with businesses in Leith will be placed with the Evening News.

2. That monthly meetings will be scheduled for attendance by tram line Project Managers, Stakeholder Relationship Manager, Communications Manager and Weber Shandwick.

#### By 22 March 2005

1. A communications work model showing the roles, responsibilities and interaction between **tie**, Weber Shandwick, Stakeholder Relationship Management and CEC will be documented and agreed by all parties.

2. A proactive draft strategy for lines 1 and 2 will be developed and presented to Michael Howell, Alex Macaulay, Isabell Reid and Barry Cross on 22 March. The strategy will specifically address:

- challenges faced over the coming year
- gathering and motivating support for tram lines from all interested parties and stakeholders
- involving and motivating the community
- · getting media on side and supportive stories printed
- showing the benefits of integration
- including interested groups such as environmental and heritage
- addressing any possible changes to route
- learning from Nottingham and Dublin communications experiences.

3. That the budget proposal for the 2005 business plan should reflect the proactive approach to be taken and should be set at:

Line 1 £146,000 Line 2 £146,000

In addition, recommendations for budgets of £24,000 for each of the DPOF and INFRACO work streams should be set and used when needed either in a proactive or reactionary manner.

#### **One Ticket Marketing Plans**

A marketing proposal for One Ticket has been drafted and passed to Stuart Lockhart and Ian Carter for thoughts and amendments prior to it going to the One Ticket Board. Additional work to secure definite costs for the options identified will be progressed over the coming weeks.

#### Communication strategy and partners

Work initially concentrated on plans for the announcement of the CC Referendum Result and the Tram Lines 1 and 2 Private Bill reports.

**tie** worked closely with CEC to ensure that we were 'on message', all plans dovetailed well and **tie's** proactive approach proved worthwhile. An internal communications plan for the result was followed within **tie**, providing information as soon as it was available for **tie** staff.

A Parliamentary briefing paper for Tram Lines 1 and 2 was approved and circulated to all MSPs prior to the hearing in parliament. **tie** comment was provided for press and CEC following the approval to move tram lines 1 and 2 on to the consideration stage.

Work has now started with CEC to plan communication of the Edinburgh Transport Strategy for the coming year. **tie**'s plans for trams, Ingliston Park and Ride launch, EARL and One Ticket will dovetail together with CEC's plans for the other city transport initiatives to ensure a seamless approach. Partnership with CEC is vital to ensure that this is achieved. The overall plan will be presented on 11 April to the Transport Edinburgh Communications meeting.

#### Events

#### Tram Drivers Lunch

A lunch for former tram drivers is planned for 4 April. Sixteen former drivers have contacted **tie** following story in the Evening News and a more recent story in Outlook. Andrew Burns will represent CEC. Alistair Gunn and Alan Brotchie from Light Rail Scotland, will also attend.

The lunch will be hosted by Michael Howell and Alex Macaulay who will be supported by the Tram Project Managers and Transdev. The event, held in the City Chambers, will include:

- a presentation on the tram plans
- photos and footage of the new generation of trams
- lunch
- an opportunity to relive memories.

Former drivers who are keen to bring old photos and memorabilia to the event will be encouraged to do so. Travel expenses, within reason, will be refunded.

Media will be invited to attend the final part of the event. We hope to secure positive press coverage for this event and future tram stories.

#### Ingliston Park & Ride Launch

Two options for the Ingliston Park & Ride launch event have been drafted, priced and passed to Lindsay Murphy for approval. A date has yet to be set for the event.

The Board is asked to note the position.

Suzanne Waugh 15 March 2005